CLIMATE RISK, SOFT INFORMATION AND CREDIT SUPPLY

2024

BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2406

Laura Álvarez-Román, Sergio Mayordomo, Carles Vergara-Alert and Xavier Vives CLIMATE RISK, SOFT INFORMATION AND CREDIT SUPPLY

CLIMATE RISK, SOFT INFORMATION AND CREDIT SUPPLY ()

Laura Álvarez-Román^(**)

BANCO DE ESPAÑA

Sergio Mayordomo (***)

BANCO DE ESPAÑA

Carles Vergara-Alert (****)

IESE BUSINESS SCHOOL

Xavier Vives (*****)

IESE BUSINESS SCHOOL

(*) We are grateful to Diana Bonfim, Kristle Cortés, Ralph De Haas, Alessandro Ferrari, José Fillat, Rogier Holtermans, Vahid Saadi, Amalia Stamate, and the conference and seminar participants at the Banco de España, IESE, IE University – Banco de España – Federal Reserve Bank of St. Louis Conference, Third CESC, AREUEA International Conference Cambridge 2023, Research Workshop of the MPC Task Force on Banking Analysis for Monetary Policy, and Second Annual Workshop of the ESCB Research Cluster on Climate Change. The views expressed are those of the authors and do not necessarily reflect those of the Banco de España or the Eurosystem. Carles Vergara-Alert acknowledges financial support from the Social Trends Institute Foundation 2022, the State Research Agency of the Spanish Ministry of Science, Innovation and Universities MCIN/AEI/10.13039/501100011033 ERDF (Grant Ref. PGC2018-097335-A-I00), and NextGenerationEU/PRTR (Grant Ref. TED2021-131238B-I00). Xavier Vives acknowledges financial support from the Spanish Ministry of Science, Innovation and Universities (Grant Ref. PID2021-123113NB-I00).

(***) Banco de España, C. de Alcalá, 48, 28014 Madrid, Spain. Email: laura.alvarezroman@bde.es. (***) Banco de España, C. de Alcalá, 48, 28014 Madrid, Spain. Email: sergio.mayordomo@bde.es.

(*****) IESE Business School, Av. Pearson 21, 08034 Barcelona, Spain. Email: cvergara@iese.edu.

(*****) IESE Business School, Av. Pearson 21, 08034 Barcelona, Spain. Email: xvives@iese.edu.

Documentos de Trabajo. N.º 2406 February 2024

https://doi.org/10.53479/36112

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2024

ISSN: 1579-8666 (on line)

Abstract

We study a model of the impact of climate risk on credit supply and test its predictions using data on all wildfires and corporate loans in Spain. Our findings reveal a significant decrease in credit following climate-driven events. This result is driven by outsider banks (large and diversified), which reduce lending significantly to firms in affected areas. By contrast, due to their access to soft information, local banks (geographically concentrated) reduce their loans to opaque affected firms to a lesser extent without increasing their risk. We also find that employment decreases in affected areas where local banks are not present.

Keywords: wildfire, asymmetric information, bank heterogeneity, firm lending.

JEL classification: Q54, G21, G32.

Resumen

En este trabajo se desarrolla un modelo del impacto del riesgo climático en la oferta crediticia y se evalúan sus predicciones según datos granulares sobre incendios forestales y préstamos corporativos en España. Los hallazgos revelan una disminución significativa del crédito de las empresas que se ven expuestas a un incendio. Este resultado se atribuye a los bancos de mayor tamaño y con mayor diversificación geográfica, que reducen de manera sustancial la concesión de crédito a las empresas en las zonas afectadas. En contraste, los bancos locales (más concentrados geográficamente), debido a su acceso a información blanda, reducen en menor medida sus préstamos a las empresas afectadas y más opacas en términos contables sin incurrir en mayores riesgos. También se muestra que el empleo disminuye en las áreas afectadas donde los bancos locales no están presentes.

Palabras clave: incendio, información asimétrica, heterogeneidad bancaria, crédito corporativo.

Códigos JEL: Q54, G21, G32.

1 Introduction

In recent years, climate-related disasters have increased in number and severity¹, escalating the challenges for the financial system. Beyond the transition risks stemming from shifts towards a lower-carbon economy, it is crucial to evaluate the financial sector's responses to the rising physical risks of climate change. Understanding these responses allows the assessment of the impact of physical risks on the economy.

The effects of physical climate risks on credit supply are unlike other credit or operational risks and, therefore, are not clear *a priori*. A decrease in bank lending after a climate-related disaster could be attributed to several factors, including, but not limited to, a decline of collateral value –due to the damage of corporate real estate assets– and the economic outlook of households and firms in the affected areas (see Garmaise and Moskowitz (2009); Hosono et al. (2016); and Gallagher and Hartley (2017)). Banks could also increase lending to secure additional recovery loans for firms within disaster-stricken zones (see Chavaz (2016); Cortés and Strahan (2017); and Koetter, Noth, and Rehbein (2020)). Overall, the determinants of banks' credit allocation strategies in response to climate physical risk when there is no perfect information need more study.

In this paper, we develop a bank lending model under climate shocks and with asymmetric access to information. There are two types of banks –local banks (geographically concentrated) and outsider banks (large and diversified)– and two types of firms –transparent and opaque–. Local banks can better monitor opaque firms because they have better access to soft information than outsider banks. The model includes shocks to the banks' deposits at the bank level and firms' productivity shocks at the firm level, as well as shocks at the economy level. It also incorporates firm-specific climate shocks, which have a different impact on the marginal returns of the loans depending on each bank's ability to monitor firms. Banks face the costs of raising external financing and choose loan amounts to maximize their expected profits subject to their balance sheet constraint. In equilibrium, local banks are more effective at absorbing the impact of climate shocks because they experience a smaller decrease in their marginal returns of lending after a climate shock compared to outsider banks.

Our findings uncover that the differential use of soft information by local and outside banks in their lending decisions may be a primary mechanism influencing bank credit allocation following a climate-related disaster. According to our model, after a climate event such as a wildfire, there is a decline in the credit amount extended to the affected firms. In addition, outsider banks cannot assess accurately the impact of physical climate risk on opaque firms and, therefore, they tend to respond by significantly reducing credit. Local

¹See Smith and Katz (2013); Hulme (2014); Zscheischler et al. (2018); Zscheischler et al. (2020); and Tebaldi et al. (2021) for evidence on the increase in frequency and intensity of climate-driven disasters around the world. See Pechony and Shindell (2010); Moritz et al. (2012); Pausas and Keeley (2021) for studies that focus on wildfires.

banks have access to soft information, which allows them to restrict their lending less to those affected firms that are opaque, without taking on more risk.

We test the model's predictions using data on firms, banks, and wildfires in Spain. We do so for three main reasons. First, we focus on Spain because we can construct a unique dataset –with monthly detailed information on all companies, all banks, and all bank-firm credit relationships– that covers a long period (i.e., 2004-2017). Second, we focus on wildfires due to the level of accuracy in ascertaining whether a firm has been impacted by a fire and the growing number of such extreme events. Recent scientific studies have shown that high-intensity wildfires are increasing in frequency and severity fostered by climate change (see Organization for Economic Cooperation and Development (2023)). For example, between 1979 and 2019, the global wildfire season increased its duration by 27% (Jones et al. (2022)).² Third, we focus on wildfires in Spain because it is one of the countries most affected by wildfires in Europe (see Costa, De Rigo, Libertà, Houston Durrant, and San-Miguel-Ayanz (2020)).³ The total area burned annually in Spain has been greater than 50,000 hectares in 13 of the last 18 years and at least 46.35% of Spanish municipalities are located in areas of high wildfire risk⁴.

Our empirical design is based on a quasi-experimental design provided by wildfires. Specifically, in our empirical analyses, we use data on all wildfires in Spain from 2004 to 2017, detailed data on financial information for an average of 769,983 firms per year, and monthly data from the Banco de España on all bank-firm relationships over $\in 6,000$ which are reported by all credit institutions operating in Spain. The combination of geolocalized data on loan-firms-banks that are merged with the geolocalized wildfires enables us to implement a precise identification strategy. Then, we identify all firms that were located within a wildfire-affected area on the date. These wildfire-treated firms are assigned a dummy variable *Fire* that takes the value 1 if the firm is located within the affected area and 0 otherwise. This area is defined as the region that includes the burn-area plus a 10-kilometer peripheral ring around it. The control group consists of those firms located within a ring outside the treatment delimitation.

Our empirical analysis reveals that fires exert a negative effect on firms' access to financing. We estimate that the amount of outstanding credit of firms affected by fire drops by about 6% more than for firms with similar characteristics and located in the same municipality, but not affected by fire. These findings are unrelated to any previous information

²Notably, the Mediterranean region stood out, experiencing one of the most significant extensions with an additional 29 days added to its wildfire season (Jones et al. (2022)). As reported by the European Forest Fire Information System (EFFIS) (see https://effis.jrc.ec.europa.eu/apps/effis.statistics/ seasonaltrend), the year 2022 marked the EU's second most severe wildfire season since 2006, with a staggering 785,605 hectares consumed by flames. This statistic only fell short of the devastating record set in 2017.

³Spain accounted for 40% of EU land lost to forest fires in 2022 and the number of days per year with high to extreme wildfire risk is increasing significantly due to higher temperatures and increasing drought (see Feyen, Ciscar, Gosling, Ibarreta, and Soria (2020))

 $^{^{4}}$ Vega (2021)

advantage banks might have concerning a firm's exposure to climate events. This is so as there is no significant difference in a firm's credit growth before a fire, whether or not the firm is affected by fire.⁵

For proper identification of the role of local banks extending more credit to firms affected by fire due to their better access to soft information, we implement an econometric specification that posits three drivers of credit changes: (i) an economy-wide shock, (ii) a firm's productivity shock, and (iii) a firm's climate shock. The approach builds on our model, which depends on whether the firm is affected by fire and the bank's ability to monitor the effect of such an event on the firm. We deal with the first driver (i.e., the economy-wide shock) through province-bank-time fixed effects. The second driver (i.e., the firm's productivity shock) is directly estimated through firm-time fixed effects, which also allow us to deal with the firm's demand for credit. The extent of the firm's climate shock –the third driver– hinges on the interaction of two factors: a binary indicator denoting the firm's exposure to the fire's impact and the proportion of bank credit within the province where the firm operates. This latter factor serves as a gauge of the bank's monitoring capabilities within that province. We posit that local banks heavily rely on qualitative insights and exhibit an enhanced aptitude for assessing firms' risk, in contrast to outsider banks.

Our econometric specification allows us to exploit the variation arising from the credit supply of banks with different exposures to the province where a specific firm affected by fire is located relative to firms with similar characteristics that are not affected by fire.⁶ We show that firms affected by fire obtain more credit from local banks (i.e., banks with a higher proportion of their credit within a given province) than from outsider banks, which contract their credit supply to a greater extent. Our results are not influenced by the existence of potential public subsidies that the firms affected by fire might have received, by the collateralization of credit, or by the firms' availability of more collateralizable assets (i.e., more tangible assets). Moreover, we argue that our findings are not driven by the existence of property insurance.⁷ Finally, we discard that the credit supply of local banks to firms affected by fire is driven by the lack of lending opportunities out of the area affected by fire.

We conduct additional robustness tests to elicit an identification strategy that isolates the effect of fires on the credit supply of local banks over and above any pattern of specialization

 $^{^{5}}$ Moreover, we perform several robustness analyses to check that our results are not driven by the definition of the area affected by fire, the time length of the analysis, and the fact that firms that belong to a business group could obtain financing from other firms within the group instead of relying on bank debt.

⁶Importantly, there are no significant differences between the affected and non-affected firms in terms of size, profitability, solvency, and their distribution across sectors, which means the effects associated with the climate event should be driven by fire itself and not by the fact that affected firms are different to non-affected firms.

⁷Firms with more tangible assets are more likely to purchase property insurance (see Zou and Adams (2008)). To remove the potential effects of property insurance, we use firm-time fixed effects in our empirical analyses. This allows us to compare the credit supply of two banks to the same firm depending on their specialization in the province where the firm is located, which enables us to abstract from the role of property insurance and the coverage of such insurance.

at the sector, province, or sector-province level and of market power.⁸ We also confirm that the new credit to affected firms is not exclusively driven by relationship lending. In fact, consistently with their superior access to soft information, local banks are more likely to establish new credit relationships. We further support the role of soft information based on the distance between the borrower and the lender. More specifically, we observe that local banks extend more credit to firms affected by fire when they are closer to the lender but there are no significant differences in the credit supply of local and outsider banks when they are more distant from the borrower.

Consistently with our model, we also find that local banks extend more loans to opaque affected firms, where monitoring using soft information is more relevant. On the contrary, both local and outsider banks change their credit supply to less opaque firms that are affected by fire very similarly. Importantly, the credit supply of local banks flows to more opaque but non-distressed firms, therefore their lending practices do not induce credit misallocation.

We further confirm the information channel analyzing banks' ex-post risk-taking based on the performance of firm-bank relationships established after a fire. This approach guarantees that loan refinancing, or ever-greening, cannot impair the interpretation of our findings. We find that there are no significant differences in the performance of the new credit relationships depending on banks' exposures to each province. Therefore, although local banks extend more credit to more opaque firms affected by fire, their portfolios of credit perform similarly to those of outsider banks. Finally, we also find that employment decreases in affected areas, but this decrease is insignificant when local banks are present

Our work contributes to two strands of the literature. First, our work adds to the body of research in banking that examines the impact of asymmetric information on bank loan issuance. We build upon the classic idea that when a bank extends a loan to a company, it gains a competitive edge due to its superior knowledge about the borrower in comparison to other financial institutions (see Hodgman (1961); Kane and Malkiel (1965); Black (1975); Fama (1985); Sharpe (1990); and Holmstrom and Tirole (1997)). Other previous studies find that geographical distance affects the collection of firms' soft information by banks (Petersen and Rajan (2002); Degryse and Ongena (2005); Agarwal and Hauswal (2010); and Liberti and Petersen (2019)). We introduce heterogeneity in both banks (i.e., local versus outsider banks) and firms (i.e., transparent versus opaque firms), as well as incorporating climate-related shocks into the analysis. Overall, we contribute to this banking literature by incorporating acute physical climate risk into the lending framework of analysis in the presence of asymmetric information as well as heterogeneity in banks and firms.

Second, we add to the recently growing literature that studies the effects of climaterelated shocks on bank lending (see Schüwer, Lambert, and Noth (2019); Brown, Gustafson, and Ivanov (2021); Kacperczyk and Peydró (2022); Nguyen, Ongena, Qi, and Sila (2022);

⁸Our results are also robust to alternative definitions of the areas that are affected by fire and to the use of alternative samples of firms to take into account the staggered nature of wildfires, which might occur in different locations during different periods.

Reghezza, Altunbas, Marques-Ibanez, d'Acri, and Spaggiari (2022); and Correa, He, Herpfer, and Lel (2023)). Our paper is closely related to the work that analyzes the role of small or local banks in sustaining credit supply after natural disasters. Chavaz (2016) documents that local banks hit by the massive hurricanes in 2005 increased mortgage lending more than banks less concentrated in those areas. He shows that local banks use loan sales to finance the mortgages originated in affected areas. Similarly, Cortés and Strahan (2017) find that multimarket small banks respond by increasing mortgage lending in the areas affected by several types of natural disasters and taking credit away from other markets. Koetter, Noth, and Rehbein (2020) document that the access to large and geographically diversified intra-group markets is what determines the emergency lending by local banks, which in their paper are part of banking groups with parent banks intermediating intra-group liquidity. Our definition of local banks diverges from that examined in these papers. In our paper, local banks are defined as banks whose activities are concentrated in a specific region and, importantly, are not part of banking groups. Consequently, their activity cannot be explained by internal capital markets considerations, and we can isolate the soft information mechanism by which local banks maintain the provision of credit. Moreover, our findings cannot be driven by securitization given that only an average of 0.6% of the outstanding amount of corporate credit was securitized by Spanish banks over our sample period. As a result, we contribute to this literature by showing that local banks' better access to soft information enables them to restrict their lending less to those profitable firms that are affected by climate shocks and are opaque.⁹. Overall, local banks are better positioned than outsider banks to provide financial support to profitable companies in the aftermath of the physical climate shocks.

The remainder of the paper is organized as follows. Section 2 presents the model. Section 3 characterizes the equilibrium of the model and displays its predictions. Section 4 describes the data and the empirical strategy that we use to test these predictions. Section 5 displays the empirical results related to local banks and how they affect the relationship between climate-driven events (i.e., wildfires) and corporate credit. Section 6 provides evidence that soft information is the main channel that drives the relationship between climate risk and credit supply. Section 7 provides the quantification of the effects in the real economy focusing on employment in the affected area. Finally, section 8 concludes.

2 Model

Consider an infinite horizon model, where risk-neutral bank b provides financing to firm f. There are two types of banks: Local (l) and outsider (o) banks, $b = \{l, o\}$. For simplicity, we assume that a bank can only lend to one firm, but firms can borrow from multiple banks.

⁹Although they do not deal with climate-related shocks, Favara and Giannetti (2017), Giannetti and Saidi (2019), Dursun-de Neef (2023), and Izadi and Saadi (2023) also demonstrate that banking structure plays a key role in banks' lending strategies after economic shocks

Therefore, the budget constraint for each bank b in period t, must equal the total amount of the loan, L_{fb}^t , to the sum of its deposits, D_b^t , and external funding, B_b^t , such as equity or bonds:

$$D_b^t + B_b^t = L_{fb}^t. (1)$$

Banks (credit supply). We assume that the marginal return on a loan L_{fb}^t decreases linearly in its size as $(\bar{r}_f - \alpha_L L_{fb}^t)$, where \bar{r}_f is a constant for each firm, α_L is a positive constant, and raising external financing is costly. We assume that banks can raise deposits to a certain limit \bar{D}_b^t and the marginal cost of raising additional external financing is $\alpha_B B_b^t$, where α_B is a positive constant. At the end of period t, the economy receives a credit supply shock in the form of a shock to the deposits as follows:

$$D_b^{t+1} = D_b^t + \delta_E + \delta_b, \tag{2}$$

where δ_E and δ_b are shocks to the economy and to the specific bank b, respectively.

Firms (credit demand and climate shocks). The economy receives a credit demand shock at the end of period t in the form of a productivity shock to firm f and a climate shock, such that the marginal revenue of a loan L_{fb}^{t+1} is given by $\bar{r}_f - \alpha_L L_{fb}^{t+1} + \eta_E + \eta_f - (1 - \kappa_{fb})\nu_f$, where η_E and η_f are the productivity shocks to the economy and the firm f, respectively, and ν_f denotes a firm-specific climate shock. This shock has a lower impact on the marginal returns of the loan amount for banks that present a higher value of the scale factor κ_{fb} , which denotes the bank's ability to monitor opaque firms. We define κ_{fb} as a constant scale factor that is both firm (f) and bank (b) specific with $1 > \kappa_{fb} > 0$. Let κ_{fb} be the product of a firm-specific factor, κ^f , and a bank-specific factor, κ^b , that is, $\kappa_{fb} = \kappa^f \cdot \kappa^b$.

Local banks versus outsider banks (monitoring). We assume that local and outsider banks monitor opaque firms in different ways. Local banks present a higher value of κ^b than outsider banks, that is, $\kappa^l > \kappa^o$. The rationale behind this assumption is that the local bank has access to soft information about firms that the outsider bank does not have.

Opaque firms versus transparent firms. There are two types of firms f: transparent firms (tr) and opaque (op), with $f = \{tr, op\}$. Transparent firms are companies that operate with a high level of openness, disclosure, and accountability. These firms maintain clear and accessible communication channels with their stakeholders, including banks. Transparency allows banks to understand how the company operates and make informed decisions based on reliable information. On the other hand, opaque firms are companies that lack transparency and operate with limited disclosure of information. Opaque firms may have restricted communication channels, limited public reporting, and a lack of openness in their operations. This lack of transparency can make it challenging for banks to fully understand the company's operations, assess its performance, and make informed lending decisions. Therefore, we assume that $\kappa^{tr} > \kappa^{op}$. Overall, notice that the key mechanism is that $\kappa_{l,op} > \kappa_{o,op}$,

that is, local banks can better monitor opaque firms.¹⁰ Globally, climate shocks are better amortized by local banks, resulting in a lower impact on marginal returns. The following table summarizes the interactions between κ^f and κ^b for the different types of firms and banks in the model, as well as the parameter assumptions related to banks' monitoring of opaque firms, $\kappa^l > \kappa^o$ (i.e., high κ^l and low κ^o) and $\kappa^{tr} > \kappa^{op}$ (i.e., high κ^{tr} and low κ^{op}):

		Firms					
		Transparent	Opaque				
Banka	Local	$\kappa_{tr,l} = \kappa^{tr} \text{ (high)} \times \kappa^{l} \text{ (high)}$	$\kappa_{op,l} = \kappa^{op} \text{ (low)} \times \kappa^l \text{ (high)}$				
DallKS	Outsider	$\kappa_{tr,o} = \kappa^{tr} \text{ (high)} \times \kappa^{o} \text{ (low)}$	$\kappa_{op,o} = \kappa^{op} (\text{low}) \times \kappa^{o} (\text{low})$				

The bank's problem. Bank b chooses the loan amounts L_{fb}^t and L_{fb}^{t+1} that lends to firm f to maximize its expected profits subject to its balance sheet constraint (1) and the process for the bank's deposits (2). Therefore, bank b solves the following problem:

$$\max_{\{L_{fb}^{t}, L_{fb}^{t+1}\}} \{ [\bar{r_{f}} - \alpha_{L} L_{fb}^{t}] \cdot L_{fb}^{t} - \alpha_{B} B_{b}^{t} + [\bar{r_{f}} + \eta_{E} + \eta_{f} - \alpha_{L} L_{fb}^{t+1} - (1 - \kappa_{fb}) \nu_{f}] \cdot L_{fb}^{t+1} - \alpha_{B} B_{b}^{t+1} \}$$
(3)

such that

$$D_{b}^{t} + B_{b}^{t} = L_{fb}^{t}$$
$$D_{b}^{t+1} + B_{b}^{t+1} = L_{fb}^{t+1}$$
$$D_{b}^{t+1} = D_{b}^{t} + \delta_{E} + \delta_{b}$$

Notice that we assume no discount for the profits at time t + 1.

3 Equilibrium and Model Predictions

By plugging the constraints into the objective function above, we obtain the following optimization problem:

$$\max_{\{L_{fb}^{t}, L_{fb}^{t+1}\}} \{ [\bar{r_{f}} - \alpha_{L} L_{fb}^{t} - \alpha_{B}] \cdot L_{fb}^{t} + \alpha_{B} D_{b}^{t} + [\bar{r_{f}} + \eta_{E} + \eta_{f} - \alpha_{L} L_{fb}^{t+1} - (1 - \kappa_{fb}) \nu_{f} - \alpha_{B}] \cdot L_{fb}^{t+1} + \alpha_{B} D_{b}^{t+1} \}$$

$$(4)$$

The first-order conditions (FOCs) of this problem can be expressed as:

$$\bar{r_f} - \alpha_B - 2\alpha_L L_{fb}^t = 0 \tag{5}$$

$$\bar{r_f} - \alpha_B - 2\alpha_L L_{fb}^{t+1} + \eta_E + \eta_f - (1 - \kappa_{fb})\nu_f = 0.$$
(6)

In equilibrium we can determine the amount of loans:

¹⁰We could even assume that $\kappa_{l,tr} = \kappa_{o,tr}$. In other words, transparent firms can be monitored equally by local and outsider banks.

$$L_{fb}^{*(t)} = (\bar{r_f} - \alpha_B)/2\alpha_L \tag{7}$$

$$L_{fb}^{*(t+1)} = (\bar{r}_f - \alpha_B + \eta_E + \eta_f - (1 - \kappa_{fb})\nu_f)/2\alpha_L.$$
(8)

Notice that any reduction in marginal revenue (i.e. raising the cost of external financing, negative economy-wide and firm-specific productivity, and climate shocks) has a negative effect on the loan amount in equilibrium. By combining (7) and (8) into a single difference equation, we obtain an expression for the change in the loan amount that firm f obtains from bank b:

$$\Delta L_{fb} = (\eta_E + \eta_f - (1 - \kappa_{fb})\nu_f)/2\alpha_L \tag{9}$$

Finally,

$$\Delta L_{fb} = \underbrace{-\frac{1 - \kappa_{fb}}{2\alpha_L}\nu_f}_{\text{Firm's climate shock}} + \underbrace{\frac{1}{2\alpha_L}\eta_E}_{\text{Economy-wide shocks}} + \underbrace{\frac{1}{2\alpha_L}\eta_f}_{\text{Firm's productivity shock}}$$
(10)

where we obtain that ΔL_{fb} depends on the climate shock that affects the firm, as well as the economic-wide and firm's productivity shocks.

In the remainder of this section, we develop the main testable implications of the model, which we organize into two propositions that lead to 5 testable hypotheses. First, we focus on the study of the effects of climate shocks on changes in loan amounts in general. Proposition 1 formalizes this first model implication.

Proposition 1. A climate shock reduces the amount of credit lent to a given firm.

Proof. The derivative of (10) with respect to ν_f is given by the following expression, which is negative because $0 < \kappa_{fb} < 1$ and $\alpha_L > 0$.

$$\frac{\partial \Delta L_{fb}}{\partial \nu_f} = -\frac{1 - \kappa_{fb}}{2\alpha_L} < 0. \tag{11}$$

In summary, this proposition shows that climate-related events lead to a significant decline in corporate loans. Given that this derivative is negative for all banks, no matter their ability to monitor firms, this proposition shows that climate-related events lead to a decline in firms' credit. Following Proposition 1, we propose a hypothesis about the expected impact of climate shocks on loan amounts, as follows:

Hypothesis 1: If a firm is affected by a climate event, then its amount of loan debt declines after the climate event.

Second, we study whether banks' access to soft information affects the loan reduction driven by climate shocks. We specifically analyze the change in loan amounts lent by local banks (i.e., banks that perform good monitoring at the local level) when compared to the change in loan amounts lent by outsider banks (i.e., banks that perform bad monitoring at the local level) in the event of a climate shock, with direct implications of the firms' opaqueness on the loan reduction. The emphasis is placed on the change in loan amounts lent by local banks to more opaque firms when compared to the change in loan amounts lent by outsider banks. Proposition 2 formalizes this second equilibrium implication of the model.

Proposition 2. The reduction of the loan amount lent from bank b to firm f decreases with κ_{fb} .

Proof. Equation (11) (i.e., the derivative of (10) with respect to ν_f) becomes less negative as the value of κ_{fb} increases, where $0 < \kappa_{fb} < 1$.

Under our assumptions, the reduction in loans for opaque firms is lower for local banks –when compared to outsider banks– because local banks show a greater value of $\bar{\kappa_b}$ than outsider banks, that is, $\bar{\kappa_l} > \bar{\kappa_o}$, which results in $\kappa_{l,op} > \kappa_{o,op}$. Overall, local banks reduce credit less than outsider banks.

From the second proposition, we derive the following two testable hypotheses related to the role of information on the causal relationship between climate risk and firms' credit:

Hypothesis 2: Local banks reduce lending to firms to a significantly lesser extent than outsider banks.

The main implication of this hypothesis is that in areas where local banks are not present and firms get loans only from outsider banks, the impact of a climate shock is larger. In other words, the decline in business loans after a climate event is driven by outsider banks. In addition, if our results are due to better soft information, local banks should extend more credit to opaque firms.

Hypothesis 3: Local banks lend significantly more to opaque firms than outsider banks.

Next, we check that the mechanism that drives these results is access to soft information and not risk-taking. Particularly, we want to confirm that local banks do not increase their risk by lending significantly more or lending more to opaque firms than outsider banks. Hypothesis 4 summarizes this prediction, which has important implications for banking stability (see Blickle, Hamerling, and Morgan (2021); Noth and Schüwer (2023); and Klomp (2014) for studies on the effect of natural disasters on financial stability).

Hypothesis 4: Local banks do not take more risk after a climate shock.

Finally, we study the impact of local banks in the local economy in the presence of climate shocks.¹¹ Building upon the fact that there is a lower contraction of credit associated with local banks after a climate event, we expect that employment in areas where local banks are present does not decrease after a climate shock as it does in areas with no presence of local banks. Hypothesis 5 predicts that local banks play a critical role in mitigating the effects of climate shocks in the real economy.

¹¹Previous research has examined the impact of bank lending frictions on employment using firm-level data from the 2008 Financial Crisis (see Chodorow-Reich (2014)).

Hypothesis 5: Employment in fire-affected areas served by local banks does not decrease significantly after fire.

4 Data and Empirical Strategy

In this section, we explain the data that we use in our empirical analysis and define the empirical strategy that we implement to test the main theoretical predictions from the model in section 3.

4.1 Data

We assemble data from multiple sources to create a dataset that contains information on all firms' characteristics, corporate loans, and attributes of all lender banks, as well as all wildfires in Spain for the period 2004-2017. Our complete micro approach provides an effective way to test the effects of physical climate risk on credit supply based on geolocalized matches at the loan- firm-, and bank- levels.

Firms' characteristics. We use the Banco de España's Central Balance Sheet Data Office (CBSDO). It contains the balance sheets and profit and loss accounts, as well as other non-financial characteristics such as industry, year of incorporation, and demographic status, among others, for an average of 769,983 non-financial corporations per year with adequate accounting quality (based on the Banco de España's internal classification criteria).¹² We merge CBSDO with the Iberian Balance Sheet Analysis System (SABI) from which we obtain the geographical coordinates where each firm is located. We exclude from our sample firms that belong to the agriculture, livestock, forestry, and fishing sectors as the location of their economic activity in many cases differs from where the firms are domiciled.¹³

Corporate loan data. We use the Banco de España's Central Credit Register (CCR) data. It contains monthly information on all bank-firm relationships over a reporting threshold of $\leq 6,000$ for credit institutions operating in Spain.¹⁴ As loans to companies are normally larger than the reporting threshold, we can claim that we have the whole population of loans to those firms. We match firm information in CBSDO to all their entities' relationships by using the firm fiscal identifier, which uniquely identifies firms in all datasets.

Firms affected by wildfires. We use detailed Civio data on all fires in Spain with a burned area of at least 1 hectare from 2001 to 2017. This data contains data attributes such

 $^{^{12}}$ We provide the details about the filters that we apply to the data in the Appendix.

¹³For example, some companies in our sample that are involved in agriculture have their crops located far away from where the company is registered.

¹⁴We note that most of the external financing obtained by the firms in our sample is in the form of standard loans. Arce, Mayordomo, and Gimeno (2021) document that only 94 Spanish non-financial companies issued a bond at any time between 2006 and 2015. Moreover, the securitization of corporate loans is very low (on a monthly average basis, around 0.6% of the amount outstanding of corporate credit was securitized by Spanish banks over our sample period).

as coordinates, burned area, and time to fire extinction.¹⁵ We use the fire data from 2004 since from this year most fires are geolocated with exact precision.¹⁶ We drop fires with a lack of precise geolocation. Moreover, we restrict the sample to those fires equal to or larger than 500 hectares, which corresponds to the threshold that determines whether a fire can be considered a large fire with a sizeable economic impact.¹⁷ Panel A of Table 1 shows the descriptive statistics of fires in our sample. The initial sample consists of 54,032 fires equal to or greater than 1 hectare with a mean size of 24.6 hectares burned. There are 337 fires larger or equal to 500 hectares,¹⁸ which represents 48.7% of the total burned area between 2004 and 2017.

[INSERT TABLE 1 AROUND HERE]

4.2 Empirical Strategy

Our empirical analysis is based on a quasi-experimental design provided by wildfires. We combine the panel of geolocalized database on loans-firms-banks with the database on wild-fires such that we achieve a very precise identification strategy. Recent literature has quantified the effects of climate-driven events by using approaches that employ less granular information such as ZIP codes, counties, or municipalities as an identification method (see Ramos and Sanz (2020); Rehbein and Ongena (2022) and Ouazad and Kahn (2022)).¹⁹

We estimate the effects of climate-driven events (i.e., wildfires) on firms' loans using an OLS regression which is defined based on the model predictions. More specifically, equation (10) consists of three drivers of credit growth: (i) firm's climate shock, (ii) economy-wide shock, and (iii) firm's productivity shock. First, the firm's climate shock depends on whether the firm is affected by the wildfire event and the bank's ability to monitor the effect of such an event on the firm. The bank's monitoring ability is defined based on the credit granted in the province where the firm is located over the total amount of credit granted by that bank in Spain. We assume that local banks (i.e., those with a high proportion of their credit in one province) rely on soft information to a higher extent and exhibit a greater ability to monitor firms' risk. As a consequence, the credit contraction for a firm affected by the wildfire event

¹⁵Civio extracts data from the General Forest Fire Statistics (Estadística General de Incendios Forestales, EGIF) published by the Spanish Ministry for the Ecological Transition and the Demographic Challenge. See https://datos.civio.es/dataset/todos-los-incendios-forestales/ for more details. EGIF merges the data of all the wildfire reports from all the Spanish regions. These reports contain 150 variables. They are prepared *in situ* by fire engineers.

¹⁶Data available in EGIF covers the period 1983-2015. Wildfire data for 2016 and 2017 are incomplete.

¹⁷For more details on big fires see section 4 in López-Santalla and López-Garcia (2019).

¹⁸The biggest fire in our sample, which took place in Cortes de Pallás in 2012, has a total burnt area of 28,879 hectares.

¹⁹Issler, Stanton, Vergara-Alert, and Wallace (2023) also uses granular data but their analysis focuses on the effects of wildfires on housing and mortgage characteristics rather than firms' lending from banks.

that operates in provinces with local banks is expected to be lower than that in provinces without local banks. Second, each bank transmits differently the economy-wide shock to the firms in a given province. Therefore, the intensity of the productivity shock to the economy depends on the existence of banks specialized in lending to firms in that province (Giannetti and Saidi (2019)). Finally, the firm's productivity shock can be directly estimated through firm fixed effects.

Henceforth, we propose the following empirical specification to estimate equation (10) and disentangle the credit supply provided by local banks from the one provided by outsider banks:

$$\Delta L_{f,b,t+1} = \beta Local Bank_{b,p,t-1} \times Fire_{f,t} + \gamma_{b,p,t} + \gamma_{f,t} + \epsilon_{f,b,t+1}, \tag{12}$$

where the dependent variable, $\Delta L_{f,b,t+1}$, is the logarithm change in the amount of firm f's outstanding loans with bank b between December of year t-1 and December of year t+1.²⁰

The explanatory variable of interest is the interaction between the fraction of bank b's credit balance in province p where firm f is located, as of December of year t - 1, $LocalBank_{b,p,t-1}$, and the dummy variable $Fire_{f,t}$, which takes the value of 1 if the firm is located within the *affected* area and 0 otherwise. The coefficient β of this interaction term captures the supply of credit to firms affected by fire depending on the presence of local banks (i.e., depending on the concentration of credit of the bank in the province where the firm is located).

We consider that a firm is affected by fire if it is located within a circle defined by the *affected* or treatment area. We define this area as the region that includes the burn-area (i.e., circle with radius r) plus a 10-kilometer (10km) peripheral ring around it (see Figure 1).²¹ This ring enables us to account for the fact that some firms are not physically located in the burn-area, but are largely suffering the consequences of wildfire, for example, because of supply-chain disruption, damaged infrastructure, and utility service disruption.²² The non-affected or control area is defined as the peripheral ring (i.e., the grey ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. This is to guarantee that the group of non-affected firms is not contaminated by firms surrounding the affected area whose businesses could be ultimately damaged because of the fire. Panel B of Table 1 shows the

²⁰We adjust credit balances to deal with bank mergers so that we capture the true variation of the credit of a firm with each specific bank. In particular, before calculating the credit variation between a year t - 1and a year t + 1, we aggregate the credit balances of a firm by the entities that will operate —due to bank mergers— as a single entity by the end of year t + 1. Therefore, the change at the entity level is clean of the effects due to mergers of banks and reflects the real change of credit of a firm with a particular bank.

²¹We use the coordinates of the origin of the wildfire as the center of the circle and its radius, r, is such that the total area of the circle is equal to the number of hectares burned by fire. We assume a circular shape for the area burnt by each fire.

²²Appendix A2 provides a further discussion about the choice of 10 kilometers as a distance of large economic influence of a wildfire.

number of firms affected and non-affected by fires between 2004 and 2017. Importantly, there are no significant differences between the affected and non-affected firms in terms of size – measured as total assets–, returns –ROA–, solvency –capital over total assets–, and their distribution across sectors. Thus, the effects associated with the climate event should be driven by the fire itself and not by the fact that affected firms are different from non-affected firms.

[INSERT FIGURE 1 AROUND HERE]

Additionally, we use bank-province-time fixed-effects, $\gamma_{b,p,t}$, to account for concurrent bank-specific time-varying factors that affect banks' credit supply around the occurrence of the fire in a specific province p. The specification also includes firm-time fixed effects, $\gamma_{f,t}$, to proxy the firm's productivity shock. They also allow us to deal with the firm's demand for credit. As a result, we can exploit the variation arising from the credit supply of banks with different exposures to the province where a specific firm affected by fire is located relative to firms that are not affected by fire. Finally, $\epsilon_{f,b,t+1}$ in equation (12) denotes the error term.

5 Local Banks and Change in Credit After a Climate-Driven Event

In this section, we test the first two main theoretical predictions from the model. In Section 5.1, we provide the test of Hypothesis 1 and study whether the credit of firms affected by a climate event declines. In Section 5.2, we empirically test Hypothesis 2 and analyze the change in loan amounts lent by local banks when compared to the change in credit supplied by outsider banks in the event of a fire.

5.1 Change in Credit After a Climate-Driven Event

We first study whether the occurrence of a wildfire in year t is associated with a significant decrease in firms' credit (Hypothesis 1). To this aim, we estimate a variation of equation (12) in which the credit growth is defined at the firm level instead of at the bank-firm level. More specifically, we exploit the exogenous variation of firms' exposure to economic damage due to wildfires each year and run an OLS regression in which the dependent variable is the logarithm change in the amount of credit of firm f drawn between December of year t - 1and December of year t + 1, $\Delta L_{f,t+1}$. The credit growth at the firm level is regressed on the dummy variable denoting if the firm f is affected by fire at time t and a set of additional explanatory variables that include firm characteristics as of December of year t - 1 to control for its size (logarithm of total assets), profitability (return on assets), and solvency (equity over total assets). In addition, we use industry-municipality-size-time fixed effects.²³

The results of this analysis are reported in Table 2. We find that the amount of outstanding credit of firms affected by fire drops by about 6% more than for firms with similar characteristics and located in the same municipality, but not affected by fire (column 1). This result confirms that fires exert a negative effect on firms' access to financing. To check whether this effect is exclusively driven by firms that are closer to the burn area, we split the dummy variable denoting firms affected by fire, *Fire* (10km), into two groups depending on their proximity to the burn-area: *Fire* (5km) includes firms located inside the area defined by the burn-area plus a peripheral ring of 5 km outside the burn-area, and *Fire* (5km - 10km) incorporates the peripheral ring with inner radius r + 5km and outer radius r + 10km. Column (2) of Table 2 shows that firms located in both areas suffer a larger drop in their credit growth when compared to the control group.

[INSERT TABLE 2 AROUND HERE]

One potential concern with these results is that banks might have prior information regarding firm exposure to climate events. If this is the case, banks could have incorporated their prior information advantage into their ex-ante screening process and adjusted the credit supply to such firms before the fire occurred at time t. To address this concern, we perform a robustness test in which we re-estimate the specification in (1) of Table 2 but using the change in the logarithm of credit between December of year t-3 and December of year t-1as the dependent variable. Column (3) shows that there is no significant difference in the credit growth of a firm before the fire regardless of whether the firm is affected by the fire at time t.²⁴

A second concern regarding this test is that our sample contains firms that belong to a business group. Firms within a group could be located in a different municipality or province than their headquarters. Therefore, a fire could affect a given subsidiary but not its headquarters. These firms could obtain financing from the group instead of relying on debt at the subsidiary's level. To tackle this issue, we remove firms that are part of a group and re-estimate the specification in column (1). We report the outcome of this robustness test

²³Specifically, we consider the interaction of a set of fixed effects that deal with the industry in which the firm operates (1-digit NACE), the municipality where the firm is located, several dummy variables dealing with the firm-size (micro, small, medium-sized, and large corporations) and year dummy variables. This set of fixed effects enables us to estimate whether the credit growth of two firms with similar characteristics within a given municipality differs between them when one of these firms is affected by fire and the other one is not. Comparing firms within a given municipality is important to deal with political connections at the municipality level or the effectiveness of forestry brigades, among other factors.

 $^{^{24}}$ We conduct this analysis using data for the period 2006–2017. We replicate the estimation from column (1) using this shorter period and find a similar effect in sign and magnitude.

in column (4) and find that the overall results presented in column (1) remain unaffected by this concern.

Moreover, we run a robustness check on the period of the dependent variable, $\Delta L_{f,t+1}$, which considers the variation of credit between December of t-1 and December of t+1 to evaluate the effects of a fire at time t. To further confirm the validity of our results, we reestimate our main specification when the dependent variable is obtained using the variation of credit between December of year t-1 and December of year t. Column (5) displays the result of this analysis, which presents a larger magnitude than our baseline outcome in column (1).

Finally, we check whether the drop in the amount of credit obtained by firms affected by fire is because some of these firms became inactive. To do so, we exclude from our sample the firms that became inactive the year in which the fire occurred or the year after.²⁵ Results of this robustness test are reported in column (6) and present the same sign and magnitude as those in column (1).

Overall, this first empirical result shows that there is a significant drop in the credit balance of firms affected by fire. In the following section, we explore whether this drop in credit balance arises from supply or demand factors and to what extent local banks play a role in extending credit to firms affected by fires.

5.2 Local Banks and Credit Supply with Climate Risk

In this subsection, we test whether local banks reduce lending to firms to a significantly lesser extent than outsider banks (Hypothesis 2). This hypothesis postulates that the decline in business loans after a climate event is driven by outsider banks. First, we present the baseline results and analysis of this test (subsection 5.2.1). Second, we conduct robustness tests on this hypothesis using different sample sets (subsection 5.2.2). Third, we examine whether the lending activity of local banks is indeed influenced by the limited lending opportunities outside the fire-affected area (subsection 5.2.3). Fourth, we account for bank specialization–across industries and industries and provinces–, bank market power, and relationship lending (subsection 5.2.4).

5.2.1 Baseline analysis

First, we provide the baseline analysis of the test of Hypothesis 2. Table 3 exhibits the results of this test. Column (1) presents the estimated coefficients from equation (12). The positive and significant sign of the interaction term $LocalBank_{b,p,t-1} \times Fire_{f,t}$ shows that a firm affected by fire obtains more credit from local banks (i.e., banks with a larger proportion of its credit outstanding in the province where the firm is located).

²⁵We define inactive firms as firms that went out of business based on their status in CBSDO the year in which the fire occurred or the year after. We consider a firm as active if it was inactive at time t but became active at time t + 1.

[INSERT TABLE 3 AROUND HERE]

The firm-time fixed effects employed in equation (12) prevent us from estimating the coefficient for the variable $Fire_{f,t}$. Thus, in column (2) we propose a more flexible specification, which is saturated with industry-municipality-size-time fixed effects instead of firm-time fixed effects. This specification allows us to control for firm-specific shocks under the assumption that firms in a given industry (1-digit NACE industry), municipality, and size in year t are affected similarly by shocks. This means that we exploit the variation arising from the credit supply of banks with different credit concentrations in a given province to firms affected by fire in a given year that have a similar size, are located in the same municipality, and operate in the same industry. The coefficient associated with the dummy $Fire_{f,t}$ measures the credit supply to firms affected by fire by banks with zero exposure to a given province. This coefficient is negative and significant, which confirms the negative effect of the fires on the credit supply of banks that are not active in the province where the firm is located. However, the higher the fraction of credit in a given province, the lower the cut in credit supply to firms affected by fire in that province. In fact, local banks with a relatively high concentration of credit in a given province could even increase the credit balance of fire-affected firms.

Furthermore, we investigate whether additional attributes of banks, aside from the proportion of credit allocated within a specific province, lead to a significant effect on credit supply to firms affected by fire. To do so, we include different interaction terms in our baseline specification. Columns (3)-(5) in Table 3 report the results of this analysis. In column (3) we include the interaction of the logarithm of total assets, TA with the dummy $Fire_{f,t}$. The interaction term is not significant and, therefore, we reject the hypothesis that the results in columns (1) and (2) are driven by the fact that the bank's exposure to a given province reflects just the size of the bank. In column (4), we show that our results are not driven by the banks' reporting standards because the interaction between the dummy IRB, which is equal to one if the bank uses internal rating-based models and zero otherwise, and the dummy $Fire_{f,t}$ is not statistically significant. The dummy IRB deals with the reporting standards and it can be understood as a variable that measures the quality of risk management in a bank.²⁶ In column (5), we include all the previous interaction terms as well as the interaction of the bank's capital, Cap, and ROA. These results confirm that the bank characteristic that leads to a statistically significant differential effect is the fraction of credit that the bank has in a specific province.

5.2.2 Alternative samples

In this subsection, we provide robustness tests based on alternative samples. Firstly, we address the concern that wildfires might happen in a staggered way, that is, fires might

 $^{^{26}}$ Banco de España approved the IRB models for the first time in 2008. We assume that the *IRB* dummy is always equal to one for the observations linked to those banks that had adopted the IRB model as of the end of our sample period.

occur in different locations during different periods. If locations were close enough, firms subject to wildfires early in our data could appear later as controls. However, we assess the robustness of our findings by conducting an analysis that excludes from the sample firms once they are affected by fire (column 2 in Table 4) and that considers only firms the first year that they appear in the sample (column 3). They show that our results are robust to the concerns regarding the potential staggered nature of wildfires.

Secondly, one could question the exclusion of firms geographically located in a ring between the treatment and control groups from the sample (see Figure 1). To address this issue, we restrict our sample to firms located within the ring of r + 20km (i.e., radius of the fire area plus 20km) surrounding the edge of the burn-area, such that the *non-affected* or control area of this alternative sample is defined as the peripheral ring with inner radius r + 10km and outer radius r + 20km.²⁷ Results are reported in column (4). We confirm that there are significant differences between the treatment and this alternative control group but consistently with the idea that the activity of firms in this new non-affected area might be contaminated, we observe that the estimated coefficient is lower than in column (1).

[INSERT TABLE 4 AROUND HERE]

Thirdly, we respond to the concern that local banks could be aware of the firms' abilities to receive subsidies. To abstract from the effect of public subsidies or aids on credit supply, we remove from our sample those firms that have received any aid or subsidy according to the information available in the CBSDO. Results are reported in column (5) of Table 4 and confirm that subsidies do not drive local banks' credit supply.

Another factor that might contribute to explaining credit supply by local banks to firms affected by a fire is the requirement of guarantees. Collateral contributes to mitigating not only asymmetric information but also the potential losses faced by the bank in case of a firm default. Therefore, we re-estimate equation (12) using only the evolution of credit without guarantees. If the activity of a local bank is driven by a more extensive use of guarantees, then we should expect no differences between the credit supply of local and outsider banks to firms affected by fire. However, the results in column (6) fully support those obtained in the baseline analysis (column 1).

Finally, although we find that the credit supply of local banks is also channeled through loans without guarantees, banks may extend more credit to firms with more tangible assets just because in case of liquidation, the recovery rate would be higher (see Davydenko and Franks (2008)). Alternatively, firms with more tangible assets could suffer more losses in case they are affected by fire. To understand, which of the two effects dominates, we split the sample into two parts depending on whether their ratio of tangible assets over total assets is below or above the median of the distribution in each year. We find that local banks extend

²⁷Note that these firms were not considered in previous analyses to avoid including areas that are not directly affected by fire but their activity might suffer from externalities of the fire.

new credit similarly to firms no matter their ratio of tangible assets (see columns (7) and (8) in Table 4).

5.2.3 Banks' lending opportunities

Local banks' credit supply to firms affected by fire could be driven by the local banks' lack of lending opportunities out of the area affected by fire, which forces them to lend to affected firms. To check this possibility we compute the lending opportunities (LO) that banks have in a given province p in a given year t as the ratio of the sales of the firms non-affected by fire in that province over the total amount of sales of the firms in the province as in Mayordomo and Rachedi (2022):

$$LO_{p,t} = \frac{\sum_{f \in PROV_p}^{Non-affected} Sales_{f,p,t}}{\sum_{f \in PROV_p}^{All} Sales_{f,p,t}}$$
(13)

A high value of the variable $LO_{p,t}$ implies that, in a given province, there are relatively more lending opportunities outside the set of firms affected by fire. Therefore, if a bank operates in areas with fewer lending opportunities, then it might be forced to continue lending to affected firms.²⁸

To test whether the role of local banks is exclusively explained by the lack of lending opportunities –instead of the advantages of soft information–, we split the sample of firms into two groups depending on whether the province offers low or high lending opportunities to banks. We assume that a given province offers low opportunities in a given year t when the measure $LO_{p,t}$ is in the bottom quintile of the distribution across provinces and it is not the case when it is above the 20th percentile. The results for the two subsamples of firms are reported in columns (1) and (2) of Table 5. We observe that banks with a higher share in a given province lend significantly more to firms that are affected by fire for the two subsamples. However, the credit supply of local banks to affected firms is higher in provinces with lower lending opportunities. Moreover, we obtain similar results when we use alternative definitions of lending opportunities based on the gross value added (columns 3 and 4), the total assets (columns 5 and 6), and the employment (columns 7 and 8) instead of sales.

[INSERT TABLE 5 AROUND HERE]

 $^{^{28}}$ Alternatively, they can reshuffle their loan portfolio away from these provinces. For instance, local banks could be forced to lend outside their local areas and make distant loans when they face fierce competition in their local branch markets (see Granja, Leuz, and Rajan (2022)). The latter possibility is beyond the scope of our paper and we exclusively focus on the lending practices within the affected province.

5.2.4 The Role of Bank Specialization and Relationship Lending

Our identification of the credit supply channel hinges on two key assumptions. First, firms' credit demand is held constant across banks within a given province. Second, changes in credit supply do not vary systematically across firms. This second assumption is challenged by the presence of firm- and sector-specific patterns in credit supply due to bank specialization (see De Jonghe, Dewachter, Mulier, Ongena, and Schepens (2020) and Paravisini, Rappoport, and Schnabl (2023)). However, the use of bank-province-time fixed-effects enables us to identify the credit supply of local banks to firms affected by fire over and above any pattern of bank specialization at the province level. We take a further step and expand our analysis in equation (12) with bank-industry-time and bank-industry-province-time fixed effects to account for bank specialization across industries and industries and provinces, respectively. This approach allows us to obtain an identification strategy that isolates the effect of fires on the credit supply of local banks over and above any pattern of specialization at the sector level and province level or the sector-province level. The results are reported in columns (2) and (3) of Table 6 and confirm that our results are not affected by any type of specialization at the levels considered in our analyses. Column (1) is identical to column (1) of Table 3 and is included to ensure comparability.

[INSERT TABLE 6 AROUND HERE]

The set of fixed effects in column (1) enables us to account for any effect that occurs at the bank-province-time level, including the consequences of market power. Nevertheless, in column (4) we advance one step and show that the information advantage is not driven by the market share but by the bank's credit concentration in a given location. To do so, we interact the market share of each bank in each province in a given year, BankMarketShare, with the dummy variable *Fire*. We obtain results of the same sign and magnitude as the baseline ones in column (1). Importantly, this new interaction term is not statistically significant and as a consequence, we discard that our results are driven by banks' market power.

Berg and Schrader (2012) illustrate that bank relationships improve credit access following a volcanic eruption in Ecuador. We next consider whether our results are affected by bank-firm relationships. One might argue that the credit supply of local banks in the affected areas could be driven not only by their better ability to monitor local firms but also by the fact that they are relationship lenders. Therefore, local banks might aim to continue the relationship with the existing clients that are affected by fire because they might be able to extract future rents from this relationship. To address this concern, we run two additional robustness tests.

First, we add the share of total credit (drawn and undrawn) that a bank bears for each firm the year before the fire event, *Share Bank-Firm*, as well as its interaction with the dummy variable *Fire* to our baseline specification in equation (12). We restrict this analysis to firm-bank pairs with an existing relationship before the fire, so that we can compute the

growth of credit.²⁹ Results are reported in column (5) of Table 6. We observe that the coefficient *Share Bank-Firm* is positive and significant, which indicates that banks with higher exposure to a particular company tend to provide that company with a greater amount of credit. This outcome is consistent with the empirical literature on relationship lending.³⁰ However, the interaction between *Share Bank-Firm* and *Fire* is not significant, which shows that the increase of credit to affected firms is not exclusively driven by relationship lending.

Second, we study whether banks are more likely to extend credit to their existing clients when they are affected by fire because they might have incentives to engage in loan evergreening to avoid an increase in their non-performing loans (NPLs) and loan loss provisions. These incentives exclusively impact loans already in place and do not apply to new credit arrangements. Therefore, to support our soft information channel, we should observe that local banks are also lending to firms without previous relationships. To confirm if this is indeed true, we conduct a robustness analysis in which the dependent variable is a dummy that takes the value 1 if the firm had a positive balance of credit (draw and undrawn) with the bank before the fire. We restrict the sample to firms with a positive variation of credit to understand whether it is more likely that local banks establish new relationships when they are affected by fire. Results are reported in column (6) of Table 6 and show that local banks are more likely to establish new credit relationships which might be consistent with their superior use of soft information.

6 Soft Information and the Relationship Between Climate Risk and Credit Supply

In this section, we provide evidence that soft information is the main channel that drives the relationship between climate risk and credit supply. First, subsection 6.1 documents that local banks extend more credit to firms affected by a fire when they are closer to the lender. In subsection 6.2, we formally test the third main theoretical prediction from the model, that is, we test whether local banks lend more to opaque firms than outsider banks (Hypothesis 3). This result indicates that soft information plays a major role as a driver of the relationship between climate-driven events and credit supply. Finally, subsection 6.3

²⁹Our restriction to firms with bank relationships before the fire closely follows the paper by Sette and Gobbi (2015), which studies the contribution of relationship lending to credit supply during a financial crisis. This approach eliminates the bias that would result from defining the growth of credit for firms with zero initial outstanding credit. If we include such firms, then a mechanical positive association would arise between firms with zero initial outstanding credit (and, thus, a very low relationship strength with the considered bank) and the growth of credit.

³⁰For further empirical evidence on the connection between relationship lending and bank credit see Petersen and Rajan (1994), Boot (2000), Elsas (2005), Bolton, Freixas, Gambacorta, and Mistrulli (2016), Sette and Gobbi (2015), Kysucky and Norden (2016), López-Espinosa, Mayordomo, and Moreno (2017), and Banerjee, Gambacorta, and Sette (2021).

confirms the soft information channel further by testing that local banks do not take more risk after a climate-driven event, which corresponds to Hypothesis 4.

6.1 The Role of Soft Information: Distance Borrower-Lender

To further illustrate that our results are not driven by relationship lending and that local banks have better access to soft information, we follow Agarwal and Hauswal (2010) and assume that greater soft information is achieved when there is a shorter distance between the borrower and the lender. Moreover, Berger, Miller, Petersen, Rajan, and Stein (2005) provide evidence consistent with small banks being better able to collect and act on soft information than large banks.³¹ In fact, outsider banks rely on hard information to a higher extent given their more advanced technology (Liberti and Petersen (2019)).

To study the role of soft information based on the distance between the borrower and the lender, we estimate equation (12) for two subsamples of bank-firm relationships. We split the sample of firms depending on whether the distance between the lender and the borrower is in the bottom or the top quintile of the distribution of the distance of all credit relationships in our sample.³² Results are reported in columns (1) and (2) of Table 7, for the bottom and top quintiles of the distance borrower-lender distribution, respectively. We observe that local banks extend more credit to firms affected by fire when they are closer to the lender (bottom quintile, column 1). Conversely, there are no significant differences in the credit supply of local and outsider banks when they are more distant from the borrower (top quintile, column 2). That is, local banks rely to a higher extent on soft information and the proximity to the borrower enables them to collect this type of information more efficiently. Interestingly, we find that the average distance between local banks, those with more than 90% of their credit balance in the province where the firm is located, and their borrowers (6.5 km).

[INSERT TABLE 7 AROUND HERE]

6.2 The Role of Soft Information: Local Banks and Credit to Opaque Firms

If the supply of business loans by local banks is driven by their superior ability to extract and use soft information, then local banks should extend more credit to those affected firms

³¹These authors state that large banks are less willing to lend "informationally difficult" credits, such as it could be the case of firms affected by fire. Other papers documenting the local nature of the soft information in credit decisions include Petersen and Rajan (2002) and Degryse and Ongena (2005).

 $^{^{32}}$ We calculate the distance between each firm in our sample and the credit institutions with which it has credit relationships after geolocating all branches using their addresses. We do not have information on the branch that granted the credit to each firm but we assume that the credit is extended by its nearest branch in the same province where the firm is located.

that are more opaque. In this subsection, we test whether local banks lend significantly more to opaque firms than outsider banks (Hypothesis 3). To test this hypothesis, we employ the same specification that we use in column (1) of Table 3 and split the sample according to the level of firm's opacity, which is measured based on accruals.³³ Columns (3)-(8) of table 7 shows the results of this test.

The specification in column (3) of Table 7 is equivalent to the one in column (1) of Table 3, but includes only firms with information on their accruals. Columns (4) and (5) report the results for the samples of the most opaque and the least opaque firms, respectively. The most and the least opaque firms are those whose levels of accruals are in the top and bottom quintiles of the distribution of accruals of the firms in our sample, respectively.

Consistently with our model, we find that both local and outsider banks change their credit supply to transparent firms that are affected by a fire in a similar manner. However, local banks provide more loans to opaque and fire-affected firms because these banks have superior access to soft information, enabling more accurate credit risk assessment.³⁴ On the contrary, banks that rely on hard information might consider it suboptimal to conduct thorough credit risk assessments for the pool of borrowers affected by fire.³⁵

In columns (7) and (8) of Table 7 we consider an alternative measure of firms' opaqueness which is defined depending on the type of questionnaire that firms report to the Banco de España. More opaque firms are those that sent the reduced questionnaire whereas less opaque (i.e., more transparent) firms are those that sent the normal questionnaire.³⁶ Consistent with the outcomes in columns (2) and (3), we observe a significant increase in the supply of credit from local banks to more opaque firms but we do not observe significant differences for more transparent firms. Column (4) is equivalent to column (1) and it is included for comparability purposes.

³³We measure a firm's opacity as the proportion of a firm's inflows and outflows of cash that cannot be predicted accurately. Earnings can be split into cash flows and accruals. Earnings quality is a concept related to earnings persistence (the ability to predict future earnings based on the information of current earnings). Sloan (1996) shows that the persistence of accruals is much lower than that of cash flows. So, firms with extremely positive and negative accruals are considered to have poor earnings quality (Bhattacharya, Desai, and Venkataraman (2013)) that, as a consequence, damage the information that lenders can infer from the earnings. Following Leuz, Nanda, and Wysocki (2003), we measure opacity as the ratio of the accruals over the net cash flows from operating activities. The accruals are measured as the absolute value of the difference between the change of non-cash current assets and non-cash current liabilities minus depreciation and amortization. The net cash flows from operating activities are defined as the absolute value of the difference between the net operating income and our proxy for accruals. Thus, the higher the ratio, the higher a firm's opacity is.

³⁴Our paper does not consider differences in the organizational structures of local and outsider banks which might affect information production and the allocation of credit. Skrastins and Vig (2019) find that increased hierarchization in a bank induces credit rationing, reduces loan performance, and generates standardization in loan contracts. The authors relate hierarchization (typical of large diversified banks) to the loss of soft information.

 $^{^{35}}$ For a better comparison of the estimated effects, we report in the last row of Table 7 the relative economic impact which is obtained as the product of the coefficient times the average concentration of credit at the province level across provinces relative to the standard deviation of the dependent variable. We observe that the effect of the concentration of credit of a given bank in a given province on credit supply to affected firms that are opaque (2.4%) is much higher than that associated with affected but transparent firms (1.5%).

Nevertheless, if local banks extend more credit to opaque and distressed firms, these outcomes may result in the misallocation of credit. To understand whether this is the case, we estimate equation (12) for two subsamples of firms: (i) opaque firms with negative equity (i.e., distressed firms); and (ii) opaque firms with positive equity (i.e., non-distressed firms).³⁷ Results are reported in columns (2) and (3) of Table 8, respectively. Column (1) contains the results for the whole sample of opaque firms for comparability. The classification of firms depending on their level of opacity is based on accruals, as in columns (2) and (3) of Table 7.

[INSERT TABLE 8 AROUND HERE]

We find that the additional credit that local banks supply to the most opaque firms is concentrated in non-distressed firms. We obtain similar results when we define opaque firms as those that send the reduced questionnaire to the CBSDO (see columns 4–6). Overall, the fact that the credit supply of local banks flows to more opaque but non-distressed firms suggests that their lending practices do not lead to credit misallocation. This finding supports that of Bolton, Freixas, Gambacorta, and Mistrulli (2016) who show that banks that acquire soft information about firms provide loans to profitable firms in crisis times.

6.3 The Role of Soft Information: Lenders' Risk-Taking with Climate Risk

In this subsection, we explore the information channel further. We have already shown that local banks extend more credit to firms affected by a wildfire –especially if they are more opaque– given that they rely on their superior ability to incorporate soft information into their lending decisions. However, although local banks' lending practices do not point to credit misallocation *ex-ante*, the quality of local banks' loan portfolios could deteriorate after lending to firms affected by fire. In other words, local banks could be increasing *ex-post* risk-taking when increasing their exposure to affected firms. Our empirical analysis shows that this is not the case. We formally test that local banks do not take more risk after a climate shock (Hypothesis 4). To test this hypothesis, we compare the *ex-post* fraction of non-performing loans (NPL) held by locals and outsiders in areas affected by fire.

First, we consider only the firm-bank pairs featuring no credit relationship before the fire.³⁸ The main advantage of this approach is that loan refinancing or evergreening cannot impair the interpretation of our findings. We first calculate the euro amount of NPLs (i.e., doubtful, non-performing, and default loans) for all new firm-bank relationships as of December of year t+2, that is, in December of year 2 after the event. Then, we define $NPL_{b,m,i,t+2}$

³⁷Bonfim, Cerqueiro, Degryse, and Ongena (2022) define zombie firms as those with negative equity.

³⁸We do this because we cannot identify the specific loan facility that turns out to be non-performing in the post-fire period. If we were to consider the firms with a relationship with a bank before the fire, some NPLs reported afterward might be associated with lending that originated before the climate event.

as the ratio of the total amount of NPLs of each bank b associated to firms affected by fire that are domiciled in a given municipality m and operate in industry i in December of year t + 2 relative to the total outstanding credit that comes from new bank-firm relationships involving firms affected by fire in that municipality and industry.³⁹ We use $NPL_{b,m,i,t+2}$ as the dependent variable in the following specification:

$$NPL_{b,m,i,t+2} = \beta LocalBank_{b,p,t-1} + \delta X_{b,m,i,t-1} + \gamma_{i,m,t} + \gamma_{b,t} + \epsilon_{b,m,i,t+2}, \tag{14}$$

where the variable of interest is the fraction of bank b's credit balance in province p, which encompass the municipality m where its borrowers are located, as of December of year t-1, $LocalBank_{b,p,t-1}$. The vector $X_{b,m,i,t-1}$ contains the average characteristics (i.e., size, profitability, and solvency) of firms that had a positive credit exposure to bank b as of December of year t-1 in municipality m and industry i. We aim to control for the characteristics of bank b portfolio of existing borrowers that could affect bank future lending policies. $\gamma_{i,m,t}$ and $\gamma_{b,t}$ denote the use of fixed effects at the industry-municipality-time and bank-time levels.⁴⁰

Column (1) of Table 9 shows the results obtained from equation (14). We show that there is no significantly different performance of the new credits granted after fire between local and outsider banks. In other words, although local banks extend more credit to firms affected by fire, their portfolios of credit perform similarly to outsider banks' portfolios. Similar results are obtained when we estimate the most saturated specification in equation (14) as shown in column (2). Furthermore, we broadened our analysis to encompass all bank-firm pairs in areas affected by fire, irrespective of their credit relationship before the climate event. The outcomes of this analysis can be found in columns (3)-(4), which show analogous results to the ones presented in columns (1)-(2). We also find that the profitability and solvency of local banks are not damaged as compared to those of outsider banks as a consequence of these lending practices (untabulated). Overall, these results confirm that local banks manage to provide more credit without increasing their risk.

[INSERT TABLE 9 AROUND HERE]

7 Effects in the Real Economy

We have documented that there is a significant decrease in firms' credit after fire. We have also shown that local banks reduce lending to a lesser extent than outsider banks. In this subsection, we study the effect of these results on the real economy. Specifically, we test whether employment in fire-affected areas served by local banks does not decrease significantly (Hypothesis 5).

³⁹The maturity of more than 90% of the new credit granted during our sample period is lower than one year. Therefore, a 2-year time window enables us to deal with the performance of the credit granted after fire until their maturity.

⁴⁰Note that we cannot use bank-province-time fixed effects as in previous specifications because it would prevent us from estimating the coefficient associated to the variable of interest.

To estimate how local banks contribute to mitigating the negative consequences on firm real outcomes, we propose the following specification:

$$\Delta Employm_{f,t+2} = \beta Fire_{f,t} + \delta X_{f,t-1} + \gamma_{i,m,s,t} + \epsilon_{f,t+2}, \tag{15}$$

where $\Delta Employm_{f,t+2}$ denotes the growth of the average number of employees at the firm level between t-1 and year t+2 and $Fire_{f,t}$ is a dummy variable that indicates whether the firm is affected by fire at time t. Let $X_{f,t-1}$ and $\gamma_{i,m,s,t}$ denote control for firm characteristics (solvency, size, and profitability) and fixed effects at the industry-municipality-size-time level, respectively. $\epsilon_{f,t+2}$ represents the error term.

Next, we divide the sample into two groups based on whether local banks operate in the municipality where the company is situated.⁴¹ Results are reported in Table 10. Column (1) reports the results obtained for the whole sample of firms and shows that the occurrence of a fire in year t is associated with a drop in the employment of the firms in areas affected by fire two years later. Columns (2) and (3) show the results obtained for the subsample of firms in municipalities with and without local banks, respectively. Importantly, the drop in employment after fire is specific to municipalities where local banks are not active lenders. On the contrary, the employment of firms affected by fire in municipalities with local banks does not decrease as compared to the employment of non-affected firms. Columns (4) and (5) show similar results when we use a stricter definition of local banks.

[INSERT TABLE 10 AROUND HERE]

8 Conclusions

This paper shows that local banks are better positioned than outsider banks to support borrowers affected by physical climate risks. We use a simple model of bank lending under climate shocks with asymmetric access to information and test its main predictions using data on all wildfires and bank-firm credit relationships in Spain from 2004 to 2017. We find that climate-related events lead to a significant decline in corporate loans in the affected areas. This reduction is driven by outsider banks, which drastically contract lending in the affected areas. In contrast, access to and superior use of soft information enable local banks to be more accurate in their lending practices. Interestingly, local banks lend more to opaque firms than outsider banks without incurring a greater risk exposure. Finally, the paper documents that the ability of local banks to extend credit after climate-related disasters benefits the economy. Specifically, employment in fire-affected areas served by local

 $^{^{41}}$ We consider that a municipality has active local banks if any firm in the municipality has a positive credit exposure to a bank with more than 90% of its credit balance in the province where the firm is located. This split enables us to exploit the exogenous variation in the existence (or not) of local banks operating in a specific municipality.

banks does not decrease significantly after fire, while employment decreases in fire-affected areas without local banks.

The findings of this paper provide relevant policy implications. Our results suggest that local banks play a critical role in mitigating the effects of climate shocks in the real economy, mainly because they can use soft information to provide recovery lending to the opaque firms that have been impacted by climate disasters. Overall, networks of local banks have more access to the necessary soft information to buffer local climate shocks, when compared to relatively large and diversified outsider banking systems.

Figures and Tables



Figure 1: Definition of affected and non-affected firms by wildfires. This figure shows a sketch of the areas where firms have been affected and those that remain unaffected by wildfires. We consider the *affected* or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The *non-affected* or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km.

Size	Number of	Mean	Median	Std. Dev.	p5	p95	Mean time to
(hectares)	fires						extinguish (hours)
>= 1	54,032	24.6	3	276	1	59.8	12.3
>= 500	337	$1,\!923.7$	1,009.0	$2,\!896.5$	533.6	$7,\!161.0$	165.8

Panel A: Descriptive statistics of fires (2004-2017)

	Affected firms		Non affec	ted firms
	Obs.	Mean	Obs.	Mean
Log (total assets)	$54,\!317$	5.8	423,342	5.9
ROA (%)	54,317	-1.4	423,342	-1.5
Capital over total assets $(\%)$	$54,\!317$	14.7	423,342	14.4

Panel B: Descriptive statistics of firms affected and non-affected by fires (2004-2017)

	Affec	eted firms	Non affected firms			
	Obs.	Percentage	Obs.	Percentage		
Industrial	8,627	15.9	72,934	17.2		
Retail	15,726	29.0	115,722	27.3		
Services	18,469	34.0	$153,\!191$	36.2		
Construction	$11,\!495$	21.2	$81,\!495$	19.3		

Panel C: Sector distribution of firms affected and non-affected by fires (2004-2017)

Table 1: **Descriptive statistics.** Panel A of this table reports the number of fires in each group according to their size and their summary statistics. The last column shows the mean of the number of hours employed to extinguish fires. Panel B shows the number of observations and the mean of several firm characteristics: size as Log(total assets), profitability as ROA, and solvency as capital over total assets depending on whether they are affected by fire or not. Panel C displays the distribution of firms across sectors depending on whether they are affected by fire or not. We only consider fires greater or equal to 500ha. We consider the *affected* or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The *non-affected* or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 20km.

Dep. Var.: $\Delta L_{f,t+1}$	(1)	(2)	(3)	(4)	(5)	(6)
	All firms	All firms	All firms	No groups	All firms	Active firms
	[t-1, t+1]	[t-1, t+1]	[t-3, t-1]	[t-1, t+1]	[t-1, t]	[t-1, t+1]
					a secondululu	
Fire (10km)	-0.059*		-0.005	-0.054*	-0.079^{***}	-0.057*
	(0.032)		(0.035)	(0.032)	(0.025)	(0.032)
Fire (5km)		-0.065*				
		(0.038)				
Fire (5km-10km)		-0.057*				
		(0.033)				
Observations	444 779	444 779	356 621	437 454	428 961	440 743
	0.196	0.100	0.106	-0194	420,301	0 107
R-squared	0.126	0.126	0.126	0.124	0.111	0.127
Firm controls	YES	YES	YES	YES	YES	YES
IndMunicipality-Size-Time FE	YES	YES	YES	YES	YES	YES

Table 2: Corporate credit growth after fire. Column (1) of this table reports the effect of a firm being affected by fire in year t on the change in the logarithm of credit (plus one to deal with zeros) between December of year t-1 and December of year t+1. We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The *non-affected* or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. The estimation period in all columns is 2004-2017. unless otherwise specified. In column (2) we split the dummy variable denoting firms affected by fire into two groups depending on whether the firm is located within the circle comprising the radius plus 5 km or between the border of this circle and the radius plus 10 km. In column (3) the dependent variable is the change in the logarithm of credit (plus one to deal with zeros) between December of year t-3 and December of year t-1 (i.e., before the fire) over the period 2006-2017. In column (4) we remove firms that are part of a business group (e.g., subsidiaries). Column (5) is analogous to column (1) but the dependent variable is the change in the logarithm corporate credit between December of year t-1 and December of year t. Column (6) is analogous to column (1) but we exclude from the sample firms that were inactive and had gone out of business based on their status in CBSDO at time t or t+1. We consider a firm as active if it was inactive at time t but became active at time t+1. All specifications include firm controls and industry-municipality-size-time fixed effects, where the industry is measured at a 1-digit level and size refers to the four categories considered by the EC: micro, small, medium-sized, and large corporations. Standard errors in parenthesis are clustered at the firm level. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $\Delta L_{f,b,t+1}$	(1)	(2)	(3)	(4)	(5)
$LocalBank \times Fire$	0.324***	0.219***	0.290**	0.328***	0.306**
Fire	(0.069)	(0.081) -0.082**	(0.110)	(0.091)	(0.118)
TA \times Fire		(0.031)	-0.007		-0.006
IRB \times Fire			(0.013)	0.003	$(0.023) \\ 0.015$
$\operatorname{Cap} \times \operatorname{Fire}$				(0.052)	$(0.072) \\ 0.437$
$ROA \times Fire$					$(0.621) \\ -3.578$
					(8.245)
Observations	664,960	892,942	664,960	664,960	664,960
R-squared	0.441	0.146	0.441	0.441	0.441
Firm-Time FE	YES	NO	YES	YES	YES
IndMunicipality-Size-Time FE	NO	YES	NO	NO	NO
Bank-Province-Time FE	YES	YES	YES	YES	YES

Table 3: Credit supply by local banks after fire. Column (1) of this table reports the results obtained from the estimation of equation (12) in which the dependent variable is the log-change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t + 1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1in the province where the firm is located (Local Bank). We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. The regression analysis includes firm-time and bank-province-time fixed effects. In column (2) we propose a less demanding specification of equation (12) which is saturated with industry-municipality-size-time fixed effects, instead of firm-time fixed effects. Industry is measured at a 1-digit level and size refers to the four categories considered by the European Commission: micro, small, medium-sized, and large corporations. These sets of fixed effects allow for the use of the dummy variable that denotes if the firm has been affected by fire as an additional variable of interest. Columns (3) - (5) are similar to column (1) but we included additional interaction terms where the dummy Fire is interacted with the logarithm of total assets -column (3)-, a dummy that denotes if the bank uses IRB models -column (4)-, and the two previous variables plus the interactions with banks' capital and ROA –column (5). The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the province and bank level. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $\Delta L_{f,b,t+1}$	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
$LocalBank \times Fire$	0.324^{***}	0.333^{***}	0.368^{***}	0.204^{**}	0.357^{***}	0.320^{***}	0.351^{***}	0.280^{**}
	(0.069)	(0.079)	(0.085)	(0.093)	(0.077)	(0.113)	(0.100)	(0.133)
Observations	664,960	602,081	386,420	$249,\!815$	$584,\!866$	472,517	$447,\!554$	$215,\!373$
R-squared	0.441	0.437	0.452	0.449	0.450	0.465	0.428	0.494
Firm-Time FE	YES	YES	YES	YES	YES	YES	YES	YES
Bank-Province-Time FE	YES	YES	YES	YES	YES	YES	YES	YES

Table 4: Credit supply by local banks after fire. Alternative samples. This table reports the results obtained for alternative samples in equation (12). Column (1) is identical to column (1) of Table 3 and is included for comparability reasons. The dependent variable is the log change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t+1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1 in the province where the firm is located (LocalBank). We consider fires with an area burned equal to or larger than 500 ha. Columns (1) - (5) are analogous to column (1) of Table 3 but with different samples of firms. In column (2) we restrict the group of affected firms to first-time fire-damaged firms. Similarly, in column (3) we only consider firms the first year that they appear in the sample (independently of whether they were affected by fire or not). In columns (1) - (3) and (5) - (8) we consider the *affected* or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. However, in column (4) we restrict our sample to those firms located less than r + 20km from the centroid of the fire, such that non-affected firms are those beyond the threshold of r + 10kmand they are not considered in columns (1) - (3). In column (5) we restrict our sample to firms that have not received subsidies in years t and t+1. In column (6) we remove firms that had credit with any type of guarantee either the year before or after fire. Column (7) only considers firms with a ratio of tangible assets over total assets equal to or below the median ratio each year. In column (8) only those firms with a ratio of tangible assets over total assets greater than the median in each year are included in the sample. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the province and bank levels. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dep. Var.: $\Delta L_{f,b,t+1}$	Sales	Sales	GVA	GVA	Assets	Assets	Employment	Employment
	Low LO	High LO	Low LO	High LO	Low LO	High Low	Low LO	High LO
LocalBank \times Fire	0.468^{***} (0.084)	0.226^{**} (0.087)	$\begin{array}{c} 0.467^{***} \\ (0.084) \end{array}$	0.227^{**} (0.087)	$\begin{array}{c} 0.359^{***} \\ (0.072) \end{array}$	0.304^{***} (0.100)	$\begin{array}{c} 0.421^{***} \\ (0.078) \end{array}$	0.246^{**} (0.099)
Observations	119,854	545,092	116,442	548,506	122,723	$542,\!225$	130,525	534,423
R-squared	0.429	0.443	0.429	0.443	0.428	0.443	0.427	0.444
Firm-Time FE	YES	YES	YES	YES	YES	YES	YES	YES
Bank-Prov-Time FE	YES	YES	YES	YES	YES	YES	YES	YES

Table 5: Credit supply by local banks after fire depending on their lending opportunities. This table reports the results obtained from the estimation of equation (12) for alternative samples of firms depending on the lending opportunities (LO) that exist in the province where they are located. The dependent variable is the log change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t+1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1 in the province where the firm is located (LocalBank). We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. We consider that lending opportunities in a specific province in a given year are low when the ratio of the total sales of the firms non-affected by fire relative to the total amount of sales of firms within a given province (see equation 13) is in the bottom quintile of the distribution across provinces (Low LO) and it is not the case when it is above the 20th percentile (High LO). Column (1) reports the results for this subsample of firms in low lending opportunities provinces whereas column (2) contains the results for the subsample of firms located in provinces where the lack of lending opportunities is not an issue according to the total amount of sales of non-affected firms. We use alternative definitions of lending opportunities based on the gross value added -columns (3) and (4)-, total assets -columns (5) and (6)-, and employment -columns (7) and (8)- instead of sales. The regression analysis includes firm-time and bank-province-time fixed effects. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the province and bank levels. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
$LocalBank \times Fire$	0.324^{***} (0.069)	0.358^{***}	0.349^{***} (0.073)	0.308^{***}	0.216^{***} (0.056)	-0.033^{**}
Bank Market Share \times Fire	(0.000)	(0.012)	(0.010)	0.210	(0.000)	(0.010)
Share Bank-Firm \times Fire				(0.188)	0.002	
Share Bank-Firm					(0.075) 0.398^{***}	
	CC 4 0 CO	CC9 401	CF9 001	664.060	(0.055)	000 F77
Observations	664,960	663,481	653,201	664,960	495,832	208,577
R-squared	0.441	0.449	0.465	0.441	0.456	0.502
Firm-Time FE	YES	YES	YES	YES	YES	YES
Bank-Province-Time FE	YES	YES	NO	YES	YES	YES
Bank-Industry-Time FE	NO	YES	NO	NO	NO	NO
Bank-Industry-Province-Time FE	NO	NO	YES	NO	NO	NO

Table 6: The role of bank specialization and relationship lending. This table reports the results obtained for alternative specifications of equation (12). Column (1) is identical to column (1) of Table 3 and is included for comparability reasons. The dependent variable is the log change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t+1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1 in the province where the firm is located (LocalBank). We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. Columns (2)-(3) include bank-industry-time and bank-industry-province-time fixed effects, respectively. Column (4) includes the interaction of the market share of bank b in December of year t-1 in the province where the firm is located with a dummy variable that is equal to one if the firm was affected by fire in year t. The market share of bank b is calculated as the total credit (drawn and undrawn) of the bank in a province over the total credit in the province. Column (5) includes the share of total credit (drawn and undrawn) that a bank has of each firm at t-1, Share Bank-Firm, and its interaction with *Fire*. In this case, the sample is restricted to the firm-bank relationships with positive credit at t-1. In column (6) the dependent variable is a dummy that takes the value 1 if the firm had a positive balance of credit (drawn and undrawn credit) with the bank at t-1 and the sample is restricted to the firms with positive variation of credit between t-1 and t+1. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the province and bank level. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $\Delta L_{f,b,t+1}$	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Distance	Distance	All	More	Less	All	Reduced	Normal
	bottom quintile	top quintile		opaque	opaque		questionaire	questionaire
LocalBank \times Fire	0.500^{**} (0.191)	0.277 (0.285)	0.301^{***} (0.077)	0.382^{**} (0.163)	0.247 (0.196)	$\begin{array}{c} 0.324^{***} \\ (0.069) \end{array}$	$\begin{array}{c} 0.332^{***} \\ (0.064) \end{array}$	$\begin{array}{c} 0.371 \\ (0.284) \end{array}$
Observations	80,663	84,649	$590,\!683$	114,001	$117,\!853$	664,960	608,003	$55,\!445$
R-squared	0.529	0.512	0.427	0.453	0.470	0.441	0.451	0.412
Firm-Time FE	YES	YES	YES	YES	YES	YES	YES	YES
Bank-Province-Time FE	YES	YES	YES	YES	YES	YES	YES	YES
Relative Economic Effect	0.038	0.010	0.019	0.024	0.015	0.020	0.021	0.015

Table 7: Soft information and credit supply by local banks after fire. The results in this table are analogous to those in column (1) of Table 3 but the firms are split in columns (1) - (2) depending on the distance in km between each borrower and the nearest branch of each lender with which it has a credit relationship and in columns (3) - (8) depending on the firm opacity. The dependent variable is the log change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t+1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1 in the province where the firm is located (Local Bank). We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r+10km and outer radius r+20km. We consider fires with an area burned equal to or larger than 500 ha. In column (1) we consider the firm-bank pairs in which this distance is the bottom quintile of the distribution whereas in column (2) we consider the pairs in which this distance is in the top quintile. Column (3) is the same as in Table 3 but it is estimated just for firms with information on their accruals and is included for comparability reasons. Column (4) reports the results for the sample of more opaque firms whereas column (5) includes the results for the least opaque corporations. Firms used in the estimation of column (4) are those whose levels of accruals are in the top quintile of the distribution of accruals of the firms in our sample whereas those in column (5) correspond to the sample of the least opaque firms whose levels of accruals are in the first quintile of the distribution of accruals. Columns (6) - (8) are analogous to columns (3) - (5) but opaque firms are those that send the reduced questionnaire to the CBSDO. The estimation period is 2004-2017. The last row reports the relative economic impact of each variable that is obtained as the product of the coefficient times the average of the share of credit of banks across provinces relative to the standard deviation of the dependent variable. Standard errors in parenthesis are clustered at the province and bank levels. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $\Delta L_{f,b,t+1}$	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Bottom quint.	Bottom quint.	Opaque	Opaque	Opaque	Red. quest.	Red. quest.	Red. quest.
	Distressed	Non-distr.	All	Distressed	Non-distr.	All	Distressed	Non-distr.
LocalBank \times Fire	-0.025 (0.954)	$\begin{array}{c} 0.568^{***} \\ (0.204) \end{array}$	0.382^{**} (0.163)	$0.152 \\ (0.313)$	$\begin{array}{c} 0.424^{***} \\ (0.157) \end{array}$	$\begin{array}{c} 0.332^{***} \\ (0.064) \end{array}$	$0.317 \\ (0.243)$	$\begin{array}{c} 0.334^{***} \\ (0.081) \end{array}$
Observations	8,659	71,105	114,001	$12,\!351$	100,316	608,003	62,856	$543,\!307$
R-squared	0.629	0.526	0.453	0.584	0.447	0.451	0.561	0.442
Firm-Time FE	YES	YES	YES	YES	YES	YES	YES	YES
Bank-ProvTime FE	YES	YES	YES	YES	YES	YES	YES	YES

Table 8: Soft information and credit supply by local banks to distressed firms after fire. Columns (1) and (2) in this table are analogous to column (1) of Table 7 but firms are further split into two groups depending on whether they can be classified as distressed firms or not. Columns (3) - (8) are analogous to those in columns (4) and (7) of Table 7 but opaque firms are divided into two groups based on their classification as distressed firms or not. We consider that a firm is in financial distress if it has negative equity. The dependent variable is the log change in credit (plus one to deal with zeros) extended by bank b to firm f between December of year t-1 and December of year t+1. The explanatory variable of interest is the interaction of a dummy variable that is equal to one if the firm was affected by fire in year t and the fraction of credit of bank b in December of year t-1 in the province where the firm is located (LocalBank). We consider the affected or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. Columns (1) - (2) consider the firm-bank pairs in which this distance in km between each borrower and the nearest branch of each lender with which it has a credit relationship is in the bottom quintile of the distribution. In this sample, column (1) reports the results only for the distressed firms and column (2) reports the results only for the non-distressed. In columns (3) - (5) we report the results obtained for the sample of firms that are classified as opaque because their levels of accruals are in the top quintile of the distribution of accruals of the firms in our sample. Column (3) reports the results for the whole sample of opaque firms and it is equivalent to column (4) in Table 7 whereas columns (4) and (5) report the results for the sample of opaque firms that are distressed and non-distressed, respectively. Columns (6) - (8) report the results obtained for those firms that are classified as opaque because they sent the reduced questionnaire to the CBSDO. Column (6) reports the results for the whole sample of opaque firms based on this criteria and it is equivalent to column (7) in Table 7 whereas columns (7) and (8) report the results for the sample of opaque firms that are distressed and non-distressed, respectively. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the province and bank levels. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $NPL_{b,m,i,t+2}$	New cred	it relationships	All credit	relationships
	(1)	(2)	(3)	(4)
LocalBank	-0.022	-0.049	-0.033	-0.158
	(0.052)	(0.515)	(0.027)	(0.123)
	· /		· · ·	~ /
Observations	$5,\!459$	5,304	20,523	20,261
R-squared	0.397	0.449	0.529	0.553
Avg Firm controls	YES	YES	YES	YES
Ind-Municipality-Time FE	YES	YES	YES	YES
Bank-Time FE	YES	YES	YES	YES
Bank-Province FE	NO	YES	NO	YES

Table 9: The quality deterioration of loans granted after fire by local banks. This table reports the results obtained from the estimation on equation (14) in which the dependent variable is the ratio of the total amount of NPLs of each bank b associated to firms affected by fire that are domiciled in a given municipality m and operate in industry i in December of year t + 2 relative to the total outstanding credit involving firms affected by fire in that municipality and industry. In columns (1) and (2) we define this ratio using only the firm-bank pairs featuring no credit relationship before the fire. The variable of interest is the fraction of bank b's credit balance in province p, which encompass the municipality m where its borrowers are located, as of December of year t-1 (LocalBank). Column (2) also includes bank-province fixed effects. Columns (3) - (4) are analogous to columns (1) - (2) but the dependent variable considers all firm-bank pairs and not only those featuring no credit relationship before the fire. All specifications include the average characteristics (size, profitability, and solvency) of the firms to which each bank had a positive credit exposure as of December of year t-1 in municipality m and industry i to control for the characteristics of the bank's portfolio of existing borrowers and industry-municipality-time and bank-time fixed-effects. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the bank level. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

Dep. Var.: $\Delta Employm_{f,t+2}$	(1)	(2)	(3)	(4)	(5)
	All	Local	Outsider	Local	Outsider
		banks	\mathbf{banks}	banks	\mathbf{banks}
		>90%	>90%	>95%	>95%
Fire	-0.013*	-0.006	-0.018*	-0.003	-0.019**
	(0.007)	(0.012)	(0.010)	(0.012)	(0.009)
Observations	466,455	206,297	260,158	176,260	290,195
R-squared	0.136	0.119	0.150	0.110	0.152
Firm controls	YES	YES	YES	YES	YES
IndMunicipality-Size-Time FE	YES	YES	YES	YES	YES

Table 10: The contribution of local banks to mitigate the negative consequences of fires on firms' employment. This table reports the effect of a firm being *affected* by fire in year t on the growth of the average number of employees between December of year t-1 and December of year t+2. We consider the *affected* or treatment area as the region that includes the burn area (i.e., a circle with radius r) plus a 10-kilometer (10km) peripheral ring around it. The non-affected or control area is defined as the peripheral ring with inner radius r + 20km and outer radius r + 40km. We exclude the firms located in the peripheral ring with inner radius r + 10km and outer radius r + 20km. We consider fires with an area burned equal to or larger than 500 ha. Column (1) reports the results for the whole sample whereas columns (2) and (4) report the results for those municipalities with active local banks whereas columns (3) and (5) contain the results obtained from municipalities without active local banks. In columns (2) and (3) we consider that a municipality has active local banks if any firm in the municipality has a positive credit exposure to a bank with more than 90% of its credit balance in the province where the firm is located whereas this threshold is set at 95% in columns (4) and (5). All specifications include firm controls and industry-municipality-size-time fixed effects, where the industry is measured at a 1-digit level and size refers to four categories: micro, small, medium-sized, and large corporations. The estimation period is 2004-2017. Standard errors in parenthesis are clustered at the firm level. ***, ** and * indicate statistical significance at the 1%, 5% and 10% levels, respectively.

References

- Agarwal, Sumit, and Robert Hauswal. (2010). "Distance and private information in lending". *The Review of Financial Studies*, 23, pp. 2757-2788. https://doi.org/10.1093/rfs/hhq001
- Arce, Óscar, Sergio Mayordomo and Ricardo Gimeno. (2021). "Making room for the needy: The credit-reallocation effects of the ECB's Corporate QE". *Review of Finance*, 25, pp. 43-84. https://doi.org/10.1093/rof/rfaa020
- Banerjee, Ryan N., Leonardo Gambacorta and Enrico Sette. (2021). "The real effects of relationship lending". *Journal of Financial Intermediation*, 48. https://doi.org/10.1016/j. jfi.2021.100923
- Berg, Gunhild, and Jan Schrader. (2012). "Access to credit, natural disasters, and relationship lending". Journal of Financial Intermediation, 21, pp. 549-568. https://doi.org/10.1016/j. jfi.2012.05.003
- Berger, Allen N., Nathan H. Miller, Mitchell A. Petersen, Raghuram G. Rajan and Jeremy C. Stein. (2005). "Does function follow organizational form? Evidence from the lending practices of large and small banks". *Journal of Financial Economics*, 76, pp. 237-269. https:// doi.org/10.1016/j.jfineco.2004.06.003
- Bhattacharya, Nilabhra, Hemang Desai and Kumar Venkataraman. (2013). "Does earnings quality affect information asymmetry? Evidence from trading costs". *Contemporary Accounting Research*, 30(2), pp. 482-516. https://doi.org/10.1111/j.1911-3846.2012.01161.x
- Black, Fischer. (1975). "Bank funds management in an efficient market". *Journal of Financial Economics*, 2(4), pp. 323-339. https://doi.org/10.1016/0304-405X(75)90008-2
- Blickle, Kristian S., Sarah N. Hamerling and Donald P. Morgan. (2021). "How bad are weather disasters for banks?". *FRB of New York Staff Report*. https://www.newyorkfed.org/ medialibrary/media/research/staff_reports/sr990.pdf?sc lang=en
- Bolton, Patrick, Xavier Freixas, Leonardo Gambacorta and Paolo Emilio Mistrulli. (2016). "Relationship and transaction lending in a crisis". *The Review of Financial Studies*, 29(10), pp. 2643-2676. https://doi.org/10.1093/rfs/hhw041
- Bonfim, Diana, Geraldo Cerqueiro, Hans Degryse and Steven Ongena. (2022). "Onsite inspecting zombie lending". *Management Science*, 69(5). https://doi.org/10.1287/ mnsc.2022.4452

- Boot, Arnoud W. A. (2000). "Relationship banking: What do we know?". *Journal of Financial Intermediation*, 9. https://doi.org/10.1006/jfin.2000.0282
- Brown, James R., Matthew T. Gustafson and Ivan T. Ivanov. (2021). "Weathering cash flow shocks". *The Journal of Finance*, 76, pp. 1731-1772. https://doi.org/10.1111/jofi.13024
- Carvalho, Vasco M., Makoto Nirei, Yukiko U. Saito and Alireza Tahbaz-Salehi. (2021). "Supply chain disruptions: Evidence from the Great East Japan Earthquake". *The Quarterly Journal of Economics*, 136(2), pp. 1255-1321. https://doi.org/10.1093/qje/qjaa044
- Chaney, Thomas, David Sraer and David Thesmar. (2012). "The collateral channel: How real estate shocks affect corporate investment". *American Economic Review*, 102(6), pp. 2381-2409. https://doi.org/10.1257/aer.102.6.2381
- Chavaz, Matthieu. (2016). "Dis-integrating credit markets: diversification, securitization, and lending in a recovery". Bank of England Working Paper, 617. https://doi.org/10.2139/ ssrn.2843683
- Chodorow-Reich, Gabriel. (2014). "The employment effects of credit market disruptions: Firm-level evidence from the 2008-9 financial crisis". *The Quarterly Journal of Economics*, 129(1), pp. 1-59. https://doi.org/10.1093/qje/qjt031
- Cochrane, Hal. (2004). "Economic loss: myth and measurement". *Disaster Prevention and Management: An International Journal*, 13(4), pp. 290-296. https://doi. org/10.1108/09653560410556500
- Correa, Ricardo, Ai He, Christoph Herpfer and Ugur Lel. (2023). "The rising tide lifts some interest rates: Climate change, natural disasters and loan pricing". European Corporate Governance Institute-Finance Working Paper, 889. https://doi.org/10.2139/ssrn.3710451
- Cortés, Kristle Romero, and Philip E. Strahan. (2017). "Tracing out capital flows: How financially integrated banks respond to natural disasters". *Journal of Financial Economics*, 125(1), pp. 182-199. https://doi.org/10.1016/j.jfineco.2017.04.011
- Costa, H., D. De Rigo, G. Libertà, T. Houston Durrant and J. San-Miguel-Ayanz. (2020). *European wildfire danger and vulnerability in a changing climate: towards integrating risk dimensions: JRC Peseta IV project: Task 9 - forest fires.* Publications Office of the European Union. https://data.europa.eu/doi/10.2760/46951
- Davydenko, Sergei A., and Julian R. Franks. (2008). "Do bankruptcy codes matter? A study of defaults in France, Germany, and the U.K." *The Journal of Finance*, 63, pp. 565-608. https://doi.org/10.1111/j.1540-6261.2008.01325.x

- De Jonghe, Olivier, Hans Dewachter, Klaas Mulier, Steven Ongena and Glenn Schepens. (2020). "Some borrowers are more equal than others: Bank funding shocks and credit reallocation". *Review of Finance*, 24(1), pp. 1-43. https://doi.org/10.1093/rof/rfy040
- Degryse, Hans, and Steven Ongena. (2005). "Distance, lending relationships, and competition". *The Journal of Finance*, 60(1), pp. 231-266. https://doi.org/10.1111/j.1540-6261.2005.00729.x
- Dursun-de Neef, H. Özlem. (2023). "Bank specialization, mortgage lending and house prices". *Journal of Banking & Finance*, 151, p. 106836. https://doi.org/10.1016/j.jbankfin.2023.106836
- Elsas, Ralf. (2005). "Empirical determinants of relationship lending". *Journal of Financial Intermediation*, 14, pp. 32-57. https://doi.org/10.1016/j.jfi.2003.11.004
- Fama, Eugene F. (1985). "What's different about banks?". *Journal of Monetary Economics*, 15(1), pp. 29-39. https://doi.org/10.1016/0304-3932(85)90051-0
- Favara, Giovanni, and Mariassunta Giannetti. (2017). "Forced asset sales and the concentration of outstanding debt: Evidence from the mortgage market". *The Journal of Finance*, 72(3), pp. 1081-1118. https://doi.org/10.1111/jofi.12494
- Feyen, L., J. Ciscar, S. Gosling, D. Ibarreta and A. Soria (eds.) (2020). *Climate change impacts and adaptation in Europe*. Publications Office of the European Union. https://data.europa.eu/doi/10.2760/171121
- Gallagher, Justin, and Daniel Hartley. (2017). "Household finance after a natural disaster: The case of hurricane Katrina". *American Economic Journal: Economic Policy*, 9(3), pp. 199-228. https://doi.org/10.1257/pol.20140273
- Garmaise, Mark J., and Tobias J. Moskowitz. (2009). "Catastrophic risk and credit markets". *The Journal of Finance*, 64(2), pp. 657-707. https://doi.org/10.1111/j.1540-6261.2009.01446.x
- Giannetti, Mariassunta, and Farzad Saidi. (2019). "Shock propagation and banking structure". *The Review of Financial Studies*, 32(7), pp. 2499-2540. https://doi.org/10.1093/rfs/hhy135
- Gordon, Peter, Harry W. Richardson and Bill Davis. (1998). "Transport-related impacts of the Northridge Earthquake". *Journal of Transportation and Statistics*, 1(2), p. 21. https://doi.org/10.21949/1404568
- Granja, João, Christian Leuz and Raghuram G. Rajan. (2022). "Going the extra mile: Distant lending and credit cycles". *Journal of Finance*, 77, pp. 1259-1324. https://doi.org/10.1111/jofi.13114

- Hallegatte, Stephane. (2015). "The indirect cost of natural disasters and an economic definition of macroeconomic resilience". World Bank Policy Research Working Paper, 7357. https://doi.org/10.1596/1813-9450-7357
- Henriet, Fanny, Stéphane Hallegatte and Lionel Tabourier. (2012). "Firm-network characteristics and economic robustness to natural disasters". *Journal of Economic Dynamics and Control*, 36(1), pp. 150-167. https://doi.org/10.1016/j.jedc.2011.10.001
- Hodgman, Donald R. (1961). "The deposit relationship and commercial bank investment behavior". *The Review of Economics and Statistics*, pp.257-268. https://doi.org/10.2307/1927288
- Holmstrom, Bengt, and Jean Tirole. (1997). "Financial intermediation, loanable funds, and the real sector". *The Quarterly Journal of Economics*, 112(3), pp. 663-691. https://www.jstor.org/stable/2951252
- Hosono, Kaoru, Daisuke Miyakawa, Taisuke Uchino, Makoto Hazama, Arito Ono, Hirofumi Uchida and Iichiro Uesugi. (2016). "Natural disasters, damage to banks, and firm investment". *International Economic Review*, 57(4), pp. 1335-1370. https://doi.org/10.1111/ iere.12200
- Hulme, Mike. (2014). "Attributing weather extremes to 'climate change' a review". *Progress in Physical Geography*, 38(4), pp. 499-511. https://doi.org/10.1177/0309133314538644
- Issler, Paulo, Richard Stanton, Carles Vergara-Alert and Nancy Wallace. (2023). "Housing and mortgage markets with climate risk: Evidence from California wildfires". UC Berkeley Working Paper. http://faculty.haas.berkeley.edu/stanton/pdf/fire.pdf
- Izadi, Mohammad, and Vahid Saadi. (2023). "Banking market structure and trade shocks". *Journal of Banking & Finance*, 153, p. 106884. https://doi.org/10.1016/j.jbankfin.2023.106884
- Jones, Matthew W., John T. Abatzoglou, Sander Veraverbeke, Niels Andela, Gitta Lasslop, Matthias Forkel, Adam J. P. Smith, Chantelle Burton, Richard A. Betts, Guido R. van der Werf, Stephen Sitch, Josep G. Canadell, Cristina Santín, Crystal Kolden, Stefan H. Doerr and Corinne Le Quéré. (2022). "Global and regional trends and drivers of fire under climate change". *Reviews of Geophysics*, 60. https://doi.org/10.1029/2020RG000726
- Kacperczyk, Marcin T., and José-Luis Peydró. (2022). "Carbon emissions and the banklending channel". Available at SSRN 3915486. https://doi.org/10.2139/ssrn.3915486
- Kane, Edward J., and Burton G. Malkiel. (1965). "Bank portfolio allocation, deposit variability, and the availability doctrine". *The Quarterly Journal of Economics*, 79(1), pp. 113-134.https:// doi.org/10.2307/1880516

- Kiel, Katherine A., and Victor A. Matheson. (2018). "The effect of natural disasters on housing prices: An examination of the Fourmile Canyon fire". *Journal of Forest Economics*, 33, pp. 1-7. https://doi.org/10.1016/j.jfe.2018.09.002
- Klomp, Jeroen. (2014). "Financial fragility and natural disasters: An empirical analysis". *Journal of Financial Stability*, 13, pp. 180-192. https://doi.org/10.1016/j.jfs.2014.06.001
- Koetter, Michael, Felix Noth and Oliver Rehbein. (2020). "Borrowers under water! Rare disasters, regional banks, and recovery lending". *Journal of Financial Intermediation*, 43, p. 100811. https://doi.org/10.1016/j.jfi.2019.01.003
- Kysucky, Vlado, and Lars Norden. (2016). "The benefits of relationship lending in a crosscountry context: A meta-analysis". *Management Science*, 62, pp. 90-110. https://www.jstor. org/stable/43834995
- Leuz, Christian, Dhananjay Nanda and Peter D. Wysocki. (2003). "Earnings management and investor protection: an international comparison". *Journal of Financial Economics*, 69(3), pp. 505-527. https://doi.org/10.1016/S0304-405X(03)00121-1
- Liberti, José M., and Mitchell A. Petersen. (2019). "Information: Hard and soft". *The Review of Corporate Finance Studies*, 8, pp. 1-41. https://doi.org/10.1093/rcfs/cfy009
- Lindell, Michael K., and Carla S. Prater. (2003). "Assessing community impacts of natural disasters". *Natural Hazards Review*, 4(4), pp. 176-185. https://doi.org/10.1061/(ASCE)1527-6988(2003)4:4(176)
- Loomis, John. (2004). "Do nearby forest fires cause a reduction in residential property values?". *Journal of Forest Economics*, 10(3), pp. 149-157. https://doi.org/10.1016/j.jfe.2004.08.001
- López-Espinosa, Germán, Sergio Mayordomo and Antonio Moreno. (2017). "When does relationship lending start to pay?". *Journal of Financial Intermediation*, 31, pp. 16-29. https:// doi.org/10.1016/j.jfi.2016.11.001
- López-Santalla, Antonio, and Marta López-García. (2019). Los Incendios Forestales en España. Decenio 2006-2015. Ministerio de Agricultura, Pesca y Alimentación, Secretaría General Técnica. https://www.miteco.gob.es/content/dam/miteco/es/biodiversidad/temas/incendiosforestales/incendios-decenio-2006-2015_tcm30-521617.pdf
- Mayordomo, Sergio, and Omar Rachedi. (2022). "The China syndrome affects banks: The credit supply channel of foreign import competition". *Journal of Financial and Quantitative Analysis*, 57(8), pp. 3114-3144. https://doi.org/10.1017/S002210902200028X

- Moritz, Max A., Marc-André Parisien, Enric Batllori, Meg A. Krawchuk, Jeff Van Dorn, David J. Ganz and Katharine Hayhoe. (2012). "Climate change and disruptions to global fire activity". *Ecosphere*, 3(6), pp. 1-22. https://doi.org/10.1890/ES11-00345.1
- Nguyen, Duc Duy, Steven Ongena, Shusen Qi and Vathunyoo Sila. (2022). "Climate change risk and the cost of mortgage credit". *Review of Finance*, 26, pp. 1509-1549. https://doi.org/10.1093/rof/rfac013
- Noth, Felix, and Ulrich Schüwer. (2023). "Natural disasters and bank stability: Evidence from the U.S. financial system". *Journal of Environmental Economics and Management*, 119, p. 102792. https://doi.org/10.1016/j.jeem.2023.102792
- Okuyama, Yasuhide, and Stephanie E. Chang. (2004). *Modeling spatial and economic impacts of disasters*. Springer Science and Business Media.
- Organization for Economic Cooperation and Development. (2023). *Taming Wildfires in the Context of Climate Change*. OECD Publishing. https://doi.org/10.1787/dd00c367-en
- Ouazad, Amine, and Matthew E. Kahn. (2022). "Mortgage finance and climate change: Securitization dynamics in the aftermath of natural disasters". *The Review of Financial Studies*, 35, pp. 3617-3665. https://doi.org/10.1093/rfs/hhab124
- Paravisini, Daniel, Veronica Rappoport and Philipp Schnabl. (2023). "Specialization in bank lending: Evidence from exporting firms". *Journal of Finance*, 78, pp. 2049-2085. https://doi. org/10.1111/jofi.13254
- Pausas, Juli G., and Jon E. Keeley. (2021). "Wildfires and global change". *Frontiers in Ecology and the Environment*, 19(7), pp. 387-395. https://doi.org/10.1002/fee.2359
- Pechony, Olga, and Drew T. Shindell. (2010). "Driving forces of global wildfires over the past millennium and the forthcoming century". *Proceedings of the National Academy of Sciences*, 107(45), pp. 19167-19170. https://doi.org/10.1073/pnas.1003669107
- Petersen, Mitchell A., and Raghuram G. Rajan. (1994). "The benefits of lending relationships: Evidence from small business data". *The Journal of Finance*, 49, pp. 3-37. https://doi. org/10.1111/j.1540-6261.1994.tb04418.x
- Petersen, Mitchell A., and Raghuram G. Rajan. (2002). "Does distance still matter? The information revolution in small business lending". *The Journal of Finance*, 57, pp. 2533-2570. https://www.jstor.org/stable/3094536
- Ramos, Roberto, and Carlos Sanz. (2020). "Backing the incumbent in difficult times: the electoral impact of wildfires". *Comparative Political Studies*, 53(3-4), pp. 469-499. https://doi.org/10.1177/0010414019858959

- Reghezza, Alessio, Yener Altunbas, David Marques-Ibanez, Costanza Rodriguez d'Acri and Martina Spaggiari. (2022). "Do banks fuel climate change?". *Journal of Financial Stability*, 62, p. 101049. https://doi.org/10.1016/j.jfs.2022.101049
- Rehbein, Oliver, and Steven Ongena. (2022). "Flooded through the back door: The role of bank capital in local shock spillovers". *Journal of Financial and Quantitative Analysis*. https://doi.org/10.1017/S0022109022000321
- Rose, Adam. (2004). "Defining and measuring economic resilience to disasters". *Disaster Prevention and Management*, 13(4), pp. 307-314. https://doi.org/10.1108/09653560410556528
- Schüwer, Ulrich, Claudia Lambert and Felix Noth. (2019). "How do banks react to catastrophic events? Evidence from hurricane Katrina". *Review of Finance*, 23, pp. 75-116. https://doi.org/10.1093/rof/rfy010
- Sette, Enrico, and Giorgio Gobbi. (2015). "Relationship lending during a financial crisis". *Journal of the European Economic Association*, 13, pp. 453-481. https://www.jstor.org/ stable/24539257
- Sharpe, Steven A. (1990). "Asymmetric information, bank lending, and implicit contracts: A stylized model of customer relationships". *The Journal of Finance*, 45(4), pp. 1069-1087. https://doi.org/10.2307/2328715
- Skrastins, Janis, and Vikrant Vig. (2019). "How organizational hierarchy affects information production". *The Review of Financial Studies*, 32, pp. 564-604. https://doi.org/10.1093/rfs/ hhy071
- Sloan, Richard G. (1996). "Do stock prices fully reflect information in accruals and cash flows about future earnings?". *Accounting Review*, pp. 289-315. https://www.jstor.org/stable/248290
- Smith, Adam B., and Richard W. Katz. (2013). "US billion-dollar weather and climate disasters: data sources, trends, accuracy and biases". *Natural Hazards*, 67(2), pp. 387-410. https://doi. org/10.1007/s11069-013-0566-5
- Stetler, Kyle M., Tyron J. Venn and David E. Calkin. (2010). "The effects of wildfire and environmental amenities on property values in northwest Montana, USA". *Journal of Forest Economics*, 69(11), pp. 2233-2243. https://doi.org/10.1016/j.ecolecon.2010.06.009
- Tebaldi, Claudia, Roshanka Ranasinghe, Michalis Vousdoukas, D. J. Rasmussen, Ben Vega-Westhoff, Ebru Kirezci, Robert E. Kopp, Ryan Sriver and Lorenzo Mentaschi. (2021). "Extreme sea levels at different global warming levels". *Nature Climate Change*, 11(9), pp. 746-751. https://doi.org/10.1038/s41558-021-01127-1

- Vega, Iker. (2021). "Half of Spain's towns at high risk of forest fires". *El País*, 13 October. https://english.elpais.com/spain/2021-10-13/half-of-spains-towns-at-highrisk-of-forest-fires.html
- Wang, Jiawei. (2023). "Natural disasters and firm leasing: A collateral channel". *Journal of Corporate Finance*, 82. https://doi.org/10.1016/j.jcorpfin.2023.102428
- Zou, Hong, and Mike B. Adams. (2008). "Debt capacity, cost of debt, and corporate insurance". *Journal of Financial and Quantitative Analysis*, 43(2), pp. 433-466. https://www.jstor.org/ stable/27647356
- Zscheischler, Jakob, Olivia Martius, Seth Westra, Emanuele Bevacqua, Colin Raymond, Radley M. Horton, Bart van den Hurk, Amir AghaKouchak, Aglaé Jézéquel, Miguel D. Mahecha, Douglas Maraun, Alexandre M. Ramos, Nina N. Ridder, Wim Thiery and Edoardo Vignotto. (2020). "A typology of compound weather and climate events". Nature Reviews Earth & Environment, 1(7), pp. 333-347. https://doi.org/10.1038/s43017-020-0060-z
- Zscheischler, Jakob, Seth Westra, Bart J. J. M. van den Hurk, Sonia I. Seneviratne, Philip J. Ward, Andy Pitman, Amir AghaKouchak, David N. Bresch, Michael Leonard, Thomas Wahl and Xuebin Zhang. (2018). "Future climate risk from compound events". *Nature Climate Change*, 8(6), pp. 469-477. https://doi.org/10.1038/s41558-018-0156-3

Appendix

A1. Data Filters

This Appendix describes extra details related to the construction of our database and, specifically, the filters that we apply. We apply several filters to the CBSDO data to define our final sample. First, we exclude firms with financial ratios that may not be comparable with those of the rest of the firms, as their goal is not profit maximization such as state-owned companies, local corporations, non-profit organizations, membership organizations, associations, foundations, and religious congregations.

Second, we also remove holding companies because their financial results may not be comparable with those of the rest of the firms. Our sample does not include foreign companies and permanent establishments of entities that do not reside in the country.

Third, financial firms and companies that do not belong to the market economy are also excluded according to the NACE industry classification. This set of firms includes financial service activities, except insurance and pension funding (64); insurance, reinsurance, and pension funding, except compulsory social security (65); activities auxiliary to financial services and insurance activities (66); public administration and defense, as well as compulsory social security (84); activities of membership organizations (94); activities of households as employers of domestic personnel (97); undifferentiated goods- and services-producing activities of private households for own use (98); and activities of extraterritorial organizations and bodies (99).

Fourth, we also remove firms that according to the CBSDO have balance sheets with (i) non-reliable monetary units or (ii) errors regarding positive/negative values. We replace the employment with missings when it is classified as non-consistent in the CBSDO.

Fifth, we remove observations that violate basic accounting rules or have impossible values (e.g. negative age).

A2. Distance of Large Economic Impact Outside the Wildfire

In this Appendix, we present a comprehensive collection of academic papers that validate the statement that businesses located within a 10-kilometer radius from the periphery of a wildfire experience adverse consequences attributable to the fire. In general, it is important to define firms' losses to understand which firms can be considered affected by a natural disaster event and what impact that event had on their businesses. The literature has extensively discussed how to account for the losses caused by natural disasters and the effect that they have on business performance (see Lindell and Prater (2003); Cochrane (2004) and Rose (2004)). As a broad explanation, losses can be distinguished between direct losses and indirect losses. Direct losses are the immediate consequences of the disaster's physical phenomenon and are suffered by businesses directly affected by the event. Destruction of a warehouse by fire or damage in a retail store by a water inundation are examples of direct losses.

Otherwise, indirect losses include all losses not caused by the natural disaster, but by its consequences and affect not only the business in the extension of the event but also business in a larger spatial scale. As defined by Okuyama and Chang (2004) indirect effects or high-order losses are "all flow losses beyond those associated with the curtailment of output as a result of hazard-induced property damage in the producing facility itself". In other words, businesses not affected but near the event might be unable to produce at pre-event levels.

Some of the causes could be explained by supply-chain disruption, damaged transportation infrastructure, utility service disruption, or damaged infrastructure (see Gordon, Richardson, and Davis (1998); Hallegatte (2015) and Carvalho, Nirei, Saito, and Tahbaz-Salehi (2021)). The main origin of underperformance is caused through the collateral constraint channel as loss of asset value by the depreciation of prices in the closest reduces the company debt and investment capacity (see Chaney, Sraer, and Thesmar (2012); Kiel and Matheson (2018) and Wang (2023)).

The literature supports the idea that a first ring embodying all firms in a 10 km radius from the edge of the wildfire is a good approximation for those companies only suffering indirect effects, especially for a devaluation on collateral assets such as properties. Stetler, Venn, and Calkin (2010) studies how prices of non-burned houses change when a big wildfire occurs near them. They show that house prices drop by -13.7% and -7.6% if they were within 5 km of the fire or between 5 km and 10km, respectively. However, they do not find significant effects beyond a 10 km distance. Other papers found similar results on prices of properties nearby (see Loomis (2004);Henriet, Hallegatte, and Tabourier (2012)).

To determine whether a firm has been affected by fire we calculate the distance in kilometers from the location of the firm to the coordinates of the fire. For this purpose, we use the Stata module geodist, which calculates geographical distances by measuring the length of the shortest path between two points along the surface of a mathematical model of the Earth. We drop fires with impossible coordinates or coordinates outside the geographical limits of Spain. Also, we drop the fire-firm observations where the fire took place previous to the existence of the firm.

Firms not affected by fire are those located in the peripheral ring with inner radius r + 20km and outer radius r + 40km. Note that we exclude from our sample those firms situated in the peripheral ring with inner radius r + 10km and outer radius r + 20km. This is to guarantee that the unaffected group of firms remains untainted by businesses in the vicinity of the affected area, which might potentially suffer damage due to the wildfire. We subject this assumption to a series of robustness tests and find that our results remain resilient in the face of the definition of the control area.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 2231 DANIEL SANTABÁRBARA and MARTA SUÁREZ-VARELA: Carbon pricing and inflation volatility.
- 2232 MARINA DIAKONOVA, LUIS MOLINA, HANNES MUELLER, JAVIER J. PÉREZ and CRISTOPHER RAUH: The information content of conflict, social unrest and policy uncertainty measures for macroeconomic forecasting.
- 2233 JULIAN DI GIOVANNI, MANUEL GARCÍA-SANTANA, PRIIT JEENAS, ENRIQUE MORAL-BENITO and JOSEP PIJOAN-MAS: Buy Big or Buy Small? Procurement Policies, Firms' Financing and the Macroeconomy*.
- 2234 PETER PAZ: Bank capitalization heterogeneity and monetary policy.
- 2235 ERIK ANDRES-ESCAYOLA, CORINNA GHIRELLI, LUIS MOLINA, JAVIER J. PÉREZ and ELENA VIDAL: Using newspapers for textual indicators: which and how many?
- 2236 MARÍA ALEJANDRA AMADO: Macroprudential FX regulations: sacrificing small firms for stability?
- 2237 LUIS GUIROLA and GONZALO RIVERO: Polarization contaminates the link with partisan and independent institutions: evidence from 138 cabinet shifts.
- 2238 MIGUEL DURO, GERMÁN LÓPEZ-ESPINOSA, SERGIO MAYORDOMO, GAIZKA ORMAZABAL and MARÍA RODRÍGUEZ-MORENO: Enforcing mandatory reporting on private firms: the role of banks.
- 2239 LUIS J. ÁLVAREZ and FLORENS ODENDAHL: Data outliers and Bayesian VARs in the Euro Area.
- 2240 CARLOS MORENO PÉREZ and MARCO MINOZZO: "Making text talk": The minutes of the Central Bank of Brazil and the real economy.
- 2241 JULIO GÁLVEZ and GONZALO PAZ-PARDO: Richer earnings dynamics, consumption and portfolio choice over the life cycle.
- 2242 MARINA DIAKONOVA, CORINNA GHIRELLI, LUIS MOLINA and JAVIER J. PÉREZ: The economic impact of conflict-related and policy uncertainty shocks: the case of Russia.
- 2243 CARMEN BROTO, LUIS FERNÁNDEZ LAFUERZA and MARIYA MELNYCHUK: Do buffer requirements for European systemically important banks make them less systemic?
- 2244 GERGELY GANICS and MARÍA RODRÍGUEZ-MORENO: A house price-at-risk model to monitor the downside risk for the Spanish housing market.
- 2245 JOSÉ E. GUTIÉRREZ and LUIS FERNÁNDEZ LAFUERZA: Credit line runs and bank risk management: evidence from the disclosure of stress test results.
- 2301 MARÍA BRU MUÑOZ: The forgotten lender: the role of multilateral lenders in sovereign debt and default.
- 2302 SILVIA ALBRIZIO, BEATRIZ GONZÁLEZ and DMITRY KHAMETSHIN: A tale of two margins: monetary policy and capital misallocation.
- 2303 JUAN EQUIZA, RICARDO GIMENO, ANTONIO MORENO and CARLOS THOMAS: Evaluating central bank asset purchases in a term structure model with a forward-looking supply factor.
- 2304 PABLO BURRIEL, IVÁN KATARYNIUK, CARLOS MORENO PÉREZ and FRANCESCA VIANI: New supply bottlenecks index based on newspaper data.
- 2305 ALEJANDRO FERNÁNDEZ-CEREZO, ENRIQUE MORAL-BENITO and JAVIER QUINTANA: A production network model for the Spanish economy with an application to the impact of NGEU funds.
- 2306 MONICA MARTINEZ-BRAVO and CARLOS SANZ: Trust and accountability in times of pandemic.
- 2307 NATALIA FABRA, EDUARDO GUTIÉRREZ, AITOR LACUESTA and ROBERTO RAMOS: Do renewable energies create local jobs?
- 2308 ISABEL ARGIMÓN and IRENE ROIBÁS: Debt overhang, credit demand and financial conditions.
- 2309 JOSÉ-ELÍAS GALLEGOS: Inflation persistence, noisy information and the Phillips curve.
- 2310 ANDRÉS ALONSO-ROBISCO, JOSÉ MANUEL CARBÓ and JOSÉ MANUEL MARQUÉS: Machine Learning methods in climate finance: a systematic review.
- 2311 ALESSANDRO PERI, OMAR RACHEDI and IACOPO VAROTTO: The public investment multiplier in a production network.
- 2312 JUAN S. MORA-SANGUINETTI, JAVIER QUINTANA, ISABEL SOLER and ROK SPRUK: Sector-level economic effects of regulatory complexity: evidence from Spain.
- 2313 CORINNA GHIRELLI, ENKELEJDA HAVARI, ELENA MERONI and STEFANO VERZILLO: The long-term causal effects of winning an ERC grant.
- 2314 ALFREDO GARCÍA-HIERNAUX, MARÍA T. GONZÁLEZ-PÉREZ and DAVID E. GUERRERO: How to measure inflation volatility. A note.
- 2315 NICOLÁS ABBATE, INÉS BERNIELL, JOAQUÍN COLEFF, LUIS LAGUINGE, MARGARITA MACHELETT, MARIANA MARCHIONNI, JULIÁN PEDRAZZI and MARÍA FLORENCIA PINTO: Discrimination against gay and transgender people in Latin America: a correspondence study in the rental housing market.

- 2316 SALOMÓN GARCÍA: The amplification effects of adverse selection in mortgage credit suply.
- 2317 METTE EJRNÆS, ESTEBAN GARCÍA-MIRALLES, METTE GØRTZ and PETTER LUNDBORG: When death was postponed: the effect of HIV medication on work, savings and marriage.
- 2318 GABRIEL JIMÉNEZ, LUC LAEVEN, DAVID MARTÍNEZ-MIERA and JOSÉ-LUIS PEYDRÓ: Public guarantees and private banks' incentives: evidence from the COVID-19 crisis.
- 2319 HERVÉ LE BIHAN, DANILO LEIVA-LEÓN and MATÍAS PACCE: Underlying inflation and asymmetric risks.
- 2320 JUAN S. MORA-SANGUINETTI, LAURA HOSPIDO and ANDRÉS ATIENZA-MAESO: The numbers of equality regulation. Quantifying regulatory activity on non-discrimination and its relationship with gender gaps in the labour market.
- 2321 ANDRES ALONSO-ROBISCO and JOSÉ MANUEL CARBÓ: Analysis of CBDC Narrative of Central Banks using Large Language Models.
- 2322 STEFANIA ALBANESI, ANTÓNIO DIAS DA SILVA, JUAN F. JIMENO, ANA LAMO and ALENA WABITSCH: New technologies and jobs in Europe.
- 2323 JOSÉ E. GUTIÉRREZ: Optimal regulation of credit lines.
- 2324 MERCEDES DE LUIS, EMILIO RODRÍGUEZ and DIEGO TORRES: Machine learning applied to active fixed-income portfolio management: a Lasso logit approach.
- 2325 SELVA BAHAR BAZIKI, MARÍA J. NIETO and RIMA TURK-ARISS: Sovereign portfolio composition and bank risk: the case of European banks.
- 2326 ANGEL-IVAN MORENO and TERESA CAMINERO: Assessing the data challenges of climate-related disclosures in european banks. A text mining study.
- 2327 JULIO GÁLVEZ: Household portfolio choices under (non-)linear income risk: an empirical framework.
- 2328 NATASCHA HINTERLANG: Effects of Carbon Pricing in Germany and Spain: An Assessment with EMuSe.
- 2329 RODOLFO CAMPOS, SAMUEL PIENKNAGURA and JACOPO TIMINI: How far has globalization gone? A tale of two regions.
- 2330 NICOLÁS FORTEZA and SANDRA GARCÍA-URIBE: A Score Function to Prioritize Editing in Household Survey Data: A Machine Learning Approach.
- 2331 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Taxing consumption in unequal economies.
- 2332 ESTHER CÁCERES and MATÍAS LAMAS: Dividend Restrictions and Search for Income.
- 2333 MARGARITA MACHELETT: Gender price gaps and competition: Evidence from a correspondence study.
- 2334 ANTON NAKOV and CARLOS THOMAS: Climate-conscious monetary policy.
- 2335 RICARDO BARAHONA, STEFANO CASSELLA and KRISTY A. E. JANSEN: Do teams alleviate or exacerbate the extrapolation bias in the stock market?
- 2336 JUAN S. MORA-SANGUINETTI and ANDRÉS ATIENZA-MAESO: "Green regulation": A quantification of regulations related to renewable energy, sustainable transport, pollution and energy efficiency between 2000 and 2022.
- 2401 LAURA HOSPIDO, NAGORE IRIBERRI and MARGARITA MACHELETT: Gender gaps in financial literacy: a multi-arm RCT to break the response bias in surveys.
- 2402 RUBÉN DOMÍNGUEZ-DÍAZ, SAMUEL HURTADO and CAROLINA MENÉNDEZ: The medium-term effects of investment stimulus.
- 2403 CLODOMIRO FERREIRA, JOSÉ MIGUEL LEIVA, GALO NUÑO, ÁLVARO ORTIZ, TOMASA RODRIGO and SIRENIA VAZQUEZ: The heterogeneous impact of inflation on households' balance sheets.
- 2404 JORGE ABAD, GALO NUÑO and CARLOS THOMAS: CBDC and the operational framework of monetary policy.
- 2405 STÉPHANE BONHOMME and ANGELA DENIS: Estimating individual responses when tomorrow matters.
- 2406 LAURA ÁLVAREZ-ROMÁN, SERGIO MAYORDOMO, CARLES VERGARA-ALERT and XAVIER VIVES: Climate risk, soft information and credit supply.